

Citation: Zeitler L, André K, Alberti A, Denby Wilkes C, Soutourina J, Goldar A (2024) A genome-wide comprehensive analysis of nucleosome positioning in yeast. PLoS Comput Biol 20(1): e1011799. https://doi.org/10.1371/journal.pcbi.1011799

Editor: Shaun Mahony, Penn State University: The Pennsylvania State University, UNITED STATES

Received: June 27, 2023

Accepted: January 3, 2024

Published: January 24, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pcbi.1011799

Copyright: © 2024 Zeitler et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: During this study, no new data was generated. Codes are made available on Zenodo (DOI <u>10.5281/zenodo.8335111</u>) Processed data are available at Figshare (<u>https://</u> doi.org/10.6084/m9.figshare.25011752.v1). As **RESEARCH ARTICLE**

A genome-wide comprehensive analysis of nucleosome positioning in yeast

Leo Zeitler[®], Kévin André, Adriana Alberti, Cyril Denby Wilkes[®]*, Julie Soutourina[®]*, Arach Goldar[®]*

Université Paris-Saclay, CEA, CNRS, Institute for Integrative Biology of the Cell (I2BC), Gif-sur-Yvette, France

* cyril.denby-wilkes@cea.fr (CDW); julie.soutourina@cea.fr (JS); arach.goldar@cea.fr (AG)

Abstract

In eukaryotic cells, the one-dimensional DNA molecules need to be tightly packaged into the spatially constraining nucleus. Folding is achieved on its lowest level by wrapping the DNA around nucleosomes. Their arrangement regulates other nuclear processes, such as transcription and DNA repair. Despite strong efforts to study nucleosome positioning using Next Generation Sequencing (NGS) data, the mechanism of their collective arrangement along the gene body remains poorly understood. Here, we classify nucleosome distributions of protein-coding genes in Saccharomyces cerevisiae according to their profile similarity and analyse their differences using functional Principal Component Analysis. By decomposing the NGS signals into their main descriptive functions, we compared wild type and chromatin remodeler-deficient strains, keeping position-specific details preserved whilst considering the nucleosome arrangement as a whole. A correlation analysis with other genomic properties, such as gene size and length of the upstream Nucleosome Depleted Region (NDR), identified key factors that influence the nucleosome distribution. We reveal that the RSC chromatin remodeler—which is responsible for NDR maintenance—is indispensable for decoupling nucleosome arrangement within the gene from positioning outside, which interfere in rsc8-depleted conditions. Moreover, nucleosome profiles in chd1A strains displayed a clear correlation with RNA polymerase II presence, whereas wild type cells did not indicate a noticeable interdependence. We propose that RSC is pivotal for global nucleosome organisation, whilst Chd1 plays a key role for maintaining local arrangement.

Author summary

In baker's yeast, as in other living organisms, the support of genetic information is tightly packaged and separated from the rest of the cell. Folding is achieved on its lowest level by wrapping the DNA around molecular complexes called nucleosomes. However, the survival of the baker's yeast requires that essential cellular processes access the genetic information protected by nucleosomes. Therefore, the nucleosome arrangement along the genome should modulate the access and use of this information and ultimately the functioning of the cell. Despite strong efforts to study nucleosome profiles, the mechanism of their collective arrangement along the genome remains poorly understood. Here, we

described in Methods section all raw data can be accessed at GEO accession numbers GSE69400 and GSE73428.

Funding: This work was supported by Fondation ARC [PGA1 RF20170205342]; Comité lle-de-France - La Ligue Nationale Contre le Cancer. K.A. was supported by a PhD training contact from the French Ministry of Higher Education and Research. L.Z. was supported by a PhD training contract from the CEA NUMERICS program, which has received funding from European Union's Horizon 2020 research and innovation program under the Marie Sklodowska-Curie grant agreement No 800945. This project has received financial support from the CNRS through the MITI interdisciplinary programs. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

tackled this issue by comparing and classifying directly the nucleosome profiles along the genes (genomic regions that are known to support the necessary informations for cell functioning). This drives us to highlight that in baker's yeast the spatial organisation of nucleosomes in genes is different from other genomic regions and this difference is maintained actively by energy consuming factors. We show that the regulation and compartmentalisation of nucleosomal organisation require the concomitant actions of local and global processes.

Introduction

The eukaryotic DNA must be tightly wrapped into the spatially constraining nucleus. This is achieved in the form of chromatin, a DNA-protein complex within which the 1-dimensional DNA is condensed around histone octamers and folded to a 3-dimensional structure. To be more precise, these histone complexes are positively-charged multiprotein structures around which the DNA molecule is locally coiled, forming a linear organisation resembling the stringing together of beads. This is why the primary structure of chromatin is commonly represented by a so-called *beads-on-a-string* model. In yeast, a nucleosome refers to \approx 147 base pairs (bp) of DNA that are wrapped around four histone units. Nucleosomes are closely spaced, with an averaged centre-to-centre distance of 165 bp, leaving roughly 15 bp of linker DNA between two adjacent histone complexes. There is a consensus that phasing is highly regular within coding regions, which is interrupted by Nucleosome Depleted Regions (NDRs) between two neighbouring genes. This observation gave rise to the barrier model, which proposes that promoter-dependent properties (e.g. bound proteins or sequence composition) pose a limit for nucleosome assembly, and arrangement occurs with respect to this barrier [1, 2]. However, it is widely accepted that various factors establish and influence the genome-wide positional nucleosome landscape, including sequence composition, transcription, and chromatin remodelers [3–6]. Since the DNA molecule must bend to wrap around the histone octamer, the local nucleotide sequence naturally affects positioning. Generally speaking, GC-rich sequences are more flexible than AT-rich ones, and they are favorable to support the presence of a nucleosome [7, 8]. However, sequence-related properties might be dependent on specific motifs.

The condensed packaging also functions as regulator for various DNA-protein interactions. Most of these processes rely on chromatin remodeler complexes, which can—by consuming energy obtained from ATP hydrolysis-move, add, or evict the histone complexes to provide or inhibit direct access to the DNA sequence [9]. In yeast, chromatin organisation is maintained by four protein families, SWI/SNF, INO80, ISW, and CHD. The RSC remodeler complex of the SWI/SNF family is the only essential chromatin remodeler in Saccharomyces cerevisiae, and it is recruited to promoter regions where it is responsible for the maintenance of NDRs [10-12]. It has also been reported that the complex has an influence on nucleosome organisation in coding regions as well as supporting RNA Polymerase II (Pol II) elongation [13]. It is presumed to restore chromatin organisation after transcription [14]. However, RSC does not exhibit an impact on regular nucleosome spacing within the gene [14, 15]. Chd1—the only member of the CHD remodeler family in yeast—is associated with various transcriptionregulating functions, including initiation, elongation, and termination [16]. It has been suggested that Chd1 stabilises perturbed nucleosomes during gene expression [17]. Isw1 and Chd1 are supposed to antagonise for nucleosome spacing within the gene, with Isw1 dominating profiles along genes with larger spacing, whereas Chd1 seems to control shorter spacing [12, 18]. It has been reported that deletion of Chd1 and Isw1 only disrupt inter-nucleosome

distances and leave the +1 position unaffected [19]. Isw2 is similarly associated with regular spacing [20], and it is particularly affecting nucleosomes close to the NDR, which is presumed to regulate transcription [21]. However, the underlying mechanism for chromatin remodeling is still under debate, and a scientific consensus is missing [22–25].

Several studies showed an interdependence between nucleosome distribution and gene expression by using MNase-seq data, a Next Generation Sequencing (NGS) technique that allows the measurement of nucleosome profiles by using MNase digestion of purified chromatin [26, 27]. It has been suggested that high gene expression correlates with low nucleosome regularity [28] as well as extreme spacing (both short and long) [18]. There are contradicting results about the correlation between transcription and nucleosome phasing. Whilst [18, 29], and [30] report that transcription increases random positioning and weakened phasing, [28] show that nucleosome phasing of highly expressed genes is increased. The depletion of Pol II exhibited increased array regularity [31]. This phenomenon seems to be conserved across species, as indicated by studies using *Drosophila* [28] and mouse cell lines [32]. The outcomes indicate that gene expression can be partially explained by nucleosome positioning over the gene body. Nonetheless, the autocorrelation of MNase-seq profiles along genes revealed that nucleosomal organisation accounts for only \approx 25% of the observed transcriptional variability, even though genes with similar regularity tend to have the same level of gene expression [33]. Surprisingly, many strains deficient for chromatin remodelers seem to show only a marginal effect on transcription [18, 19]. The only exception is *rsc8*-depleted cells, which exhibit a global decrease in gene expression [12]. A clear picture between nucleosome phasing and Pol II presence is still lacking.

Different approaches have been used to categorise collective nucleosome arrangement within transcribed regions using NGS data. However, many of them rely predominantly on measurements that describe only an average over the entire profile, such as autocorrelation measurements [33] or Pearson correlation that was adapted to include coverage [34]. Pearson correlation was also used to compare nucleosome positioning of genes before and after replication [35]. Another analysis that takes into account multiple nucleosomes upstream and downstream of the NDR was presented by [14]. However, the study focused on changes with respect to the NDR, and many phenomenological descriptions are based on the application of different analysis techniques. In order to provide comparability of nucleosome positioning changes between various mutants, we aimed to use a single mathematical framework that can be applied to all strains. To our knowledge, a unifying approach assessing location-specific phasing properties along the entire nucleosome array over varying conditions has not been proposed, and a direct comparison of the effects in different remodeler-deficient strains is difficult.

In this work, we present a genome-wide analysis of collective nucleosome positioning along the gene. We define nucleosome positioning and phasing to be the positions of the MNase-seq signal peaks over an entire single nucleosome array. By clustering the MNase-seq signals of coding regions along 6–7 histone complexes into two groups using linear Pearson cross-correlation—which measures similarity of the entire nucleosome arrangement between each gene pair—we can categorise coding regions according to their likely phasing similarity imposed by chromatin remodelers. In order to interpret how profiles are classified into the two groups, we combined the clustering with an alternative data representation via functional Principal Component Analysis (fPCA). Whilst related to the conventional Principal Component Analysis (PCA), it assumes a functional relationship between positions along the profile, whereas PCA conjectures independence of every base pair along the gene. Therefore, fPCA implicitly considers spatial dependency, which is a fundamental assumption in common nucleosome phasing models like the barrier model, where nucleosomes phasing is coordinated with respect to a barrier and each other. FPCA is commonly used in time series and signal processing, and it has been used in biology for analysing crop yield [36], identifying child growth patterns [37], as well as studying genetic variation and the allelic spectrum [38]. However, it has never been applied to the spatial interdependence of nucleosome phasing to our knowledge.

The established Pearson clusters can be visually separated by considering only two fPCs, which are therefore sufficient to interpret the gene groups. Using our analysis, we can repeatedly investigate nucleosome distributions of different chromatin remodeler-mutant strains using the same framework and interpret major differences along the entire nucleosome arrangement. By relating Pearson correlation with spatial properties along the profiles, our approach refines and complements other studies that focused either on a few individual nucleosomes close to the NDR or Transcription Starting Site (TSS); or which assessed only the average correlation of the entire array (e.g. via autocorrelation). Using MNase-seq data from yeast strains deficient for different chromatin remodelers [12, 18], we reveal that Rsc8 strongly limits coordinated nucleosome arrangement to the transcribed region. It might be therefore responsible for gene-specific phasing. By measuring how the Pearson cluster separation changes between mutants using a Support Vector Machine (SVM), we identified 5 combinations of gene deletions or protein depletions which have a notable impact on phasing properties along the entire nucleosome array compared to Wild Type (WT) conditions. Measuring correlation with other nuclear processes disclosed that none of the commonly assumed factors can easily explain long-reaching nucleosome arrangement in WT strains within the gene body. However, gene deletions—in particularly mutants that contained $chd1\Delta$ —caused a strong correlation with Pol II presence. Our results indicate a new mechanistic understanding of chromatin remodelers, where Rsc8 is responsible for long-range coordination and Chd1 for local positioning of nucleosomes. All customised source code was made available on Zenodo (DOI 10. 5281/zenodo.8335111).

Results

Nucleosome profiles can be well distinguished based on their coordinated positioning in WT

In order to compare nucleosome profiles over the gene body in WT conditions, we measured the pairwise Pearson cross-correlation of the MNase-seq data produced by [12, 18] for all protein-coding regions [39] using Eq 1. The Pearson correlation index is positive when the sequencing signals of both genes tend to change towards the same direction at the same position; and it is negative when one profile is likely to increase whereas the other one decreases. Therefore, it compares similarity of the distributional shape—i.e. whether genes are apt to contain nucleosomes at similar positions—and it does not take the scaling of the sequencing data into account. The entire arrangement for each gene is treated as an entity. For both replicates, we considered 1000 bp after and 200 bp before the +1 position (= 1200 bp, approximately the average size of a gene in *Saccharomyces cerevisiae*), containing 6–7 nucleosome dyads.

These Pearson coefficients were used as a distance metric to cluster nucleosome profiles into distinct partitions using *k*-mean clustering. In a nutshell, the algorithm divides a data set of *m* observations (the MNase-seq data) into *k* groups by minimising the variance within each cluster based on a distance metric (here, the pairwise Pearson indices over all genes). Therefore, genes within a group tend to have nucleosomes at comparable positions, whereas profiles of different groups are likely to be less similar. Using a silhouette criterion measurement which compares the similarity of an object to its own cluster with the similarity to other clusters—we determined that the Pearson coefficients are most distinctly divided when k = 2 (i.e. when having two groups, Fig 1A). As the *k*-mean clustering algorithm aggregates profiles



Fig 1. Pearson clusters and fPCA considering all protein-coding genes. (A) The silhouette plot clearly indicates that the data can be best divided into two clusters, and creating more groups would only decrease the difference between each cluster. (B) and (C) display the profiles for each cluster. Large values are given in copper, low values are black, and the colour gradient in between is uniformly scaled. It is therefore a perceptually uniform representation. Both heatmaps are normalised independently, such that their respective largest value is displayed in the strongest copper hue and their lowest value in black. Unfortunately, it is difficult to quantify visually why these clusters were established. This is particularly true because the Pearson index measures only general trends in the profile, and it does not take the scaling into account. Each row represents a gene, and the x-axis shows the position along the coding region, with the +1 nucleosome defined to be at position 0 bp. The colour code represents MNase-seq amplitude, i.e. copper values show large MNase-seq signal values, whereas dark areas indicate a low amplitude. (D) The cartoon presents the hypothesised differences that could occur between the Pearson clusters. Due to the well-positioned nucleosomes and the wave-like structure of MNase-seq data, we presume that the Pearson correlation measures coordinated nucleosome positioning along the gene. If two profiles (orange and blue) are in two different clusters, this could indicate either a shift in the exact nucleosome positions (left); or a general trend in the MNase-seq signal amplitude, i.e. either increasing or decreasing (right). (E) Pearson clusters considering all genes are linearly separable with respect to their fPC scores. This indicates that two fPCs are sufficient to interpret the gene groups. We use the symmetric Jensen-Shannon (JS) distance to describe separability between the clusters along fPC1 and fPC2. The JS distance between the cluster distributions is much larger for fPC 2 than for fPC 1. Orange and blue indicate each one group, the dashed line symbolises the best linear separation using a SVM. The x-axis represents the score of the first fPC ζ^1 , the y-axis gives the score for the second fPC ζ^2 . Both axes are scaled to the same range, points outside the range (29) were included in the analysis but not plotted. (F) When analysing the effect of the major fPCs, they describe predominantly positiondependent scaling (transparent black lines, fPC 1) and collective nucleosome phasing (transparent black arrows, fPC 2). The second fPC in WT indicates an increasing or decreasing signal magnitude as a function of distance from the TSS, suggesting stronger or weaker presence (corresponding to panel D right). The mean is given as a dashed black line, a positive contribution—i.e. adding the fPC to the mean—is displayed in magenta, and a negative contribution i.e. subtracting from the mean-is shown in green. Trends over the entire array are indicated by grey arrows. When exact positions were seemingly not affected by the fPC, we marked the positions with a grey vertical bar. See Methods for more information about how the plots were produced.

https://doi.org/10.1371/journal.pcbi.1011799.g001

together with a large pairwise Pearson index, we expected that the distribution over all pairwise Pearson correlations between nucleosome arrangements from different clusters is minimal (in the following called inter-cluster correlation). We validated the significance of the obtained Pearson gene groups by creating 500 random clusters and calculating their p-value using a one-sided Kolmogorov–Smirnov (KS) test. We averaged the p-value over the KS test result over all 500 repetitions to smooth out random fluctuations (average p-value 0.0009, S1 Fig).

This shows that nucleosomal arrays can be significantly separated into two groups using linear correlation of MNase-seq data between genes (Fig 1B and 1C).

It is difficult to straightforwardly determine how the *k*-mean clustering algorithm distinguishes between these two groups; yet the interpretation of the discriminating boundary could reveal important insights about the nucleosome positioning that is presumably imposed by chromatin remodelers. As the data by Ocampo et al. [12, 18] contains several mutants, we want to identify this discriminator repeatedly with the same mathematical framework to make the results comparable. Due to the nature of the Pearson correlation index, we can make the following assumptions. As nucleosomes are commonly well positioned in budding yeast, the MNase-seq data resembles a wave-like function with one peak approximately every 200 bp. Moreover, single histone complexes cannot overlap in a single cell. The Pearson correlation measures therefore the average phasing similarity of the entire nucleosome array of two genes. Differences in similarity come either from shifts in exact positioning (i.e. well-defined peaks, Fig 1D left) or from a change in the signal amplitude (i.e. increasing or decreasing MNase-seq magnitude over the profile or at particular locations, Fig 1D right). The clusters must be separated based on either of these two trends, or possibly a combination of them.

In the following, we refer with *coordinated positioning* to the configuration of the entire nucleosome array, and consequently, to their behaviour with respect to the two separating trends of the *k*-mean clustering. Unfortunately, the Pearson coefficient measures only the average linear pairwise correlation over the entire profile, rather than taking position-dependent particularities into account. Therefore, simply extracting the boundary from the *k*-mean clusters does not explain whether the groups were established with respect to a shift or a change in amplitude (i.e. the previously determined discriminators). Instead, it is possible to investigate how the clusters distribute with respect to the data itself; or, alternatively, with respect to a different description using dimensionality reduction methods. By evaluating the major differences between the two groups of genes, we can interpret the separating clustering boundary and link it to particular properties along the nucleosome profile.

Conventional approaches apply dimensionality reductions like PCA to visually analyse clustering distributions. However, using PCA would implicitly mean that we assume independence between every position along the gene. By using the Pearson correlation measurement, we treat every profile as a single entity, which would be violated by the independence conjecture. This also contradicts the fundamental assumption of the barrier model where the positioning of earlier nucleosomes affect later phasing. Instead, we understand the arrangement as the result of a coordinated process. We assume that the MNase-seq signal along each gene can be described as a single (unknown) continuous function, which can be approximated by a mixture of a finite number of known simpler functions (so-called basis functions). In this study, we used 20 B-Splines to represent the MNase-seq data along each gene, which were subsequently averaged to a mean profile. This permitted the application of fPCA to determine the two best-characterising functional Principal Components (fPCs) that describe each nucleosome arrangement. It incorporates specific assumptions about the spatial relationship in the distribution through the basis functions, which is the crucial difference between conventional PCA and fPCA. To be more precise, the establishment of the MNase-seq distribution is understood as a stochastic process with a mean behaviour. Each considered nucleosome array can be regarded as a realisation of this stochastic process with a deviance from the expected average distribution. Instead of defining a data representation for every gene individually, fPCA determines how the mean profile needs to be transformed to approximate a particular gene. This transformation is found by combining the basis functions over all coding regions to more complex functions that are orthonormal to each other and describe the most variance along the data (i.e. the fPCs, Eq 4). These functions transform the mean by adding them to the

average profile with a gene-specific scaling factor (i.e. ζ_i^j for the *j*-th fPC of the *i*-th gene). Consequently, every nucleosome array can be also described exclusively by the factors ζ_i^j together with the respective fPCs, and we can evaluate how the two Pearson clusters distribute with respect to these factors.

Interestingly, the two clusters—which were independently obtained by classical hierarchical k-mean clustering of Pearson coefficients—are visually neatly separated by using only the first two fPCs, indicating that they are sufficient to quantify the difference between the two sets of genes (Fig 1E). In fact, the separating boundary is almost exclusively dependent on the second fPC, whilst it is seemingly independent of the first. This is slightly less clear for the B replicate, although still distinct (S2(B) Fig). Using our previous considerations about how the algorithm establishes the two clusters, we intuited that the second fPC describes coordinated nucleosome phasing along the gene body. By analysing the effect of the second fPC on the function shape, we conclude that the clusters are determined based on the downstream presence of nucleosomes (corresponding to the right cartoon in Fig 1D). We found that the first fPC largely represents amplitude scaling at a given position, as it does not influence the location of the peak (Fig 1F). The analysis shows that position-dependent amplitude scaling and coordinated arrangement are the best two independent functional descriptors for the MNase-seq data. We show and discuss the effect of the 10 major fPCs that describe most variance in S3 Fig. Despite the fact that the ratio of explained variance is not high (21.4% and 11.5% for fPC1 and fPC2, respectively), they are completely sufficient to distinguish between the Pearson correlation groups and permit an interpretation of the linear separating boundary between the clusters.

FPCA reveals size-dependent Rsc8-mediated phasing of nucleosome positions

Since the smallest genes are \approx 300 bp long, the 1000 bp window after the +1 position can contain much more than the actual length of the coding region. In order to analyse how nucleosome phasing is affected by the gene size, we repeated the fPCA considering exclusively small (\leq 1000 bp, \approx 26.7%) or large genes (>1000 bp, \approx 73.3%). Consequently, the mean as well as the two fPCs changed, whilst we kept their allocation to the previously determined Pearson clusters the same (in the following also referred to as *all-gene* clusters). If coordinated positioning is substantially affected by the length of the transcribed region, we expected that the factors ζ_i^j of the two major fPCs should exhibit a changed behaviour with respect to the linear separability. We can confirm that the linear separation is preserved for large genes, although the boundary becomes slightly sloped (S2(C) and S2(D) Fig). The fPCs for only large genes are almost identical to the *all-gene* fPCs (S5 Fig). We therefore presume that the clusters can be still largely separated by the second fPC. We also considered a possible impact of the downstream NDR by analysing exclusively very large genes (\geq 3000 bp, \approx 11.5%). Once again, the boundary was clearly visible (S2(G) and S2(H) Fig). We concluded that the MNase-seq distribution over the first 6-7 nucleosomes of all genes larger than 1000 bp can be best clustered by the collective positioning, and it can be surmised that phasing within the gene body is only negligibly affected by the downstream NDR or nucleosomes outside the 1000 bp window.

However, the neat separation between the two clusters fully vanished for small genes (Fig 2A, for replicate B S2(E) Fig). Almost all data points belong to the same group, although both are present. We want to remind that clusters were established using all coding regions, whereas the functional representation depends now exclusively on genes smaller than 1000 bp. The newly determined fPCs include overlapping positioning inside and outside the gene body due to their varying size (S4 Fig). The fact that the clusters are not separable indicates that coordinated nucleosome phasing disappears after the Transcription Termination Site (TTS), and we



Fig 2. Nucleosome phasing is strictly limited to the gene body, which is maintained by Rsc8 but antagonised by Chd1. The cluster distribution plots in panels A-C show the distribution of both gene groups with respect to the smallgene fPCs of WT, rsc8-depleted cells, and chd1A strains. Orange and blue indicate the two clusters, and the black dashed line shows the separating boundary determined by a linear SVM. The histograms present the cluster distribution with respect to each axis. Panels D-F display the transformation of the average small-gene nucleosome profile by the two major fPCs for WT, rsc8 depletion, and chd1∆, respectively. The dashed black line as well as the solid lines in magenta and green display the mean, a positive contribution of the fPC, and a negative contribution. Turquoise arrows indicate the effect on the +1, dark blue arrows on the +4, and orange arrows on the +6 position. (A) When plotting the cluster distribution with respect to small-gene fCPs in WT, the linear separability is lost. (B) The fPCs of the rsc8-depleted strain maintain the linear separability, despite the fact that the groups were established for all genes. As we interpret the Pearson clusters as similarity in positioning between genes of 1000 bp mediated by chromatin remodelers, it possibly suggests that positioning outside coding regions influences nucleosomes inside and vice versa. (C) Whilst most mutants that were rsc8 depleted could discriminate between the all-gene clusters using small-gene fPCs, this separability is lost again in rsc8-depleted chd1∆, revealing partly antagonistic roles to maintain gene-specific phasing for Rsc8 and Chd1. (D) The effect of two fPCs sheds light on why the Pearson groups are not linearly separable in WT using small-gene fPCs. The distribution of the second fPC changes its regular wave-like form to much broader peaks and valleys after the +2 nucleosome, which corresponds to approximately the size of the smallest genes in budding yeast. (E) Nucleosome positioning in rsc8-depleted conditions is clearly visible along the entire considered region, despite the included genes being smaller. This suggests that gene-specific nucleosome arrangement cannot be maintained. It is of note that the phasing also changes for the +1 nucleosome, and the NDR can be seemingly not conserved. (F) On the other hand, rsc8-depleted chd1∆ loses the regular wave-like shape of its second fPC after the +2 nucleosome to form broader peaks, indicating the presence of gene-specific nucleosome profiles as in WT conditions. All axes are scaled to the same size for each strain; shapes and amplitudes are therefore comparable (see Methods for more details).

https://doi.org/10.1371/journal.pcbi.1011799.g002

hypothesised that the arrangement is strictly limited to the gene body. Indeed, the second small-gene fPC indicates well-defined positioning only for up to the +2 nucleosome (\approx 300 bp), and the function loses quickly its frequent wave-like shape thereafter (Fig 2D). The two major fPCs for small genes are not sufficient anymore to separate the *all-gene* clusters, which are discriminated by the presence of downstream nucleosomes. To verify our hypothesis of gene-size dependent phasing, we divided the regions into small and large genes *before* performing the Pearson clustering. When considering exclusively small genes, the two Pearson groups become linearly separable again, which is—in accordance with our hypothesis—predominantly determined by the size (S6 Fig). This shows that the nucleosome arrangement is strictly limited to the gene body.

The data produced by [12, 18] contain two replicates for $chd1\Delta$, $isw1\Delta$, and $isw2\Delta$ cells as well as *rsc8*-depleted strains, together with their combinations as double, triple, and quadruple

mutants. In order to analyse how gene-size dependent nucleosome phasing alters in varying contexts, we compared the small-gene fPCs in mutant and WT conditions. Surprisingly, the separation of the *all-gene* clusters was clearly visible for the fPCs of small coding regions in *rsc8*-depleted strains (Fig 2B). Indeed, the average MNase-seq profile exhibits phased peaks along the entire 1000 bp-window (Fig 2E), and nucleosome positioning continued outside the gene boundaries (S4 Fig). The linear separability of the *all-gene* clusters using small-gene fPCs can be found in almost all mutants which are depleted of Rsc8 (S7 Fig), with the sole exception of Rsc8-depleted *chd1*\Delta strains (Fig 2C, replicate B S8(B) Fig). Here, the groups cannot be visually separated by ζ^1 and ζ^2 , and the determined fPCs resemble small-gene fPCs in WT conditions (Fig 2F, replicate B S8(D) Fig). This indicates that the gene-specific boundaries for nucleosome phasing can be re-established, and the second fPC loses its wave-like shape again after the +2 position (S4 Fig). Consequently, we hypothesise that Chd1 and Rsc8 have partially antagonistic roles for maintaining chromatin organisation that distinguishes transcribed from non-transcribed regions. Taken together, this analysis exhibits strictly constrained and Rsc8-mediated nucleosome organisation within coding regions.

Nucleosome phasing changes in remodeler mutants

We were particularly interested in how nucleosome remodeler complexes affect coordinated phasing. To remove the gene size-dependent bias from the clustering and the established fPCs, we applied the Pearson clustering to exclusively large genes (>1000 bp) for all strains and determined their two major fPCs (S2C, S2D and S5 Figs). We can confirm that the created groups for all mutants were again significant (i.e. average p-value of a KS test with 500 random partitions was lower than 5%, see S9 Fig). Interestingly, the Pearson clusters were always visually separated by using solely the first two fPCs, although some strains exhibited a larger overlap between the groups than others (S10 and S11 Figs for replicate A and B, respectively). This suggests that coordinated phasing in all mutants can be represented by considering only the two fPCs that describe the most variance, and including more fPCs is not necessary in order to interpret the discriminating function.

The respective contribution of the two major fPCs to separate the clusters varied between the strains, suggesting that fPCA is sufficiently sensitive to capture strain-dependent consequences (S10 and S11 Figs for replicate A and B, respectively). This caused the slope of the discriminating boundary to tilt. Therefore, the transformations of the mean distribution (i.e. fPCs and their factors ζ_i^j) changed for these strains. This indicates that they had not only a global effect on the average MNase-seq profile, but also caused a gene-specific disruption of the nucleosome positioning. We deemed those strains particularly important that altered the gene-specific collective behaviour of the *entire* nucleosome array with respect to the WT. To clarify this point, we provide an example for the impact of a fictional mutant that only affects the mean but not the ensemble of nucleosomes as a whole in S12 Fig. We determined the slope for all strains using a linear SVM. As aforementioned, the boundary is tilted when only considering large genes in WT conditions (Fig 3A), and the two available replicates differ slightly. The observed deviation between replicates was used as a reference for the anticipated variability in the data. By using Eq 7, we determined chromatin remodeler-deficient strains that had a sufficiently different linear boundary with respect to the WT.

We provide three different perspectives on the data. Firstly, the cluster distribution with respect to the factors ζ_i^j together with the slope highlight mutants that particularly disrupt gene-specific collective nucleosome phasing. In the following, clusters are always indicated using the colours orange and blue. Secondly, analysing the transformation of the two major fPCs of the mean unlocks an additional understanding of the variance present in the data and



Fig 3. The fPCs, their gene specific scores, and the discriminating boundary explain collective phasing and how this changes in *chd1* Δ with respect to WT conditions. The figure shows the cluster distribution with respect to ζ_{ij}^{j} the impact of the determined fPCs, and the location-specific impact of the separating boundary for WT (i.e. panels A-C) and *chd1* Δ conditions (i.e. panels D-F). Panels A and D show the fPC scores ζ_i^j of WT and *chd1* Δ strains, respectively. For the latter, the boundary slope changed notably (black dashed line). As indicated by the fCPs in panels B and E for WT and $chd1\Delta$, respectively, the functional description of the data changes. Indeed, the second fPC of $chd1\Delta$ abates quickly after the +1, with a strong effect on the effect of the +2 (grey arrows). The dashed black line as well as the solid lines in magenta and green indicate the mean, a positive contribution of the fPC, and a negative contribution, respectively. When exact positions were seemingly not affected by the fPC, we marked the positions with a grey vertical bar. General trends are given in grey arrows along the gene. The location-specific impact of the separating boundary is given in panel C for WT and panel F for $chd1\Delta$ strains. Interestingly, despite the median distributions of the clusters (blue and orange) are clearly different with respect to the +1 and +2 in WT conditions, later positions are much more important for allocating a profile to a particular group (grey areas, mean in black). Whilst this is also true for $chd1\Delta$, the importance of later nucleosomes is even more accentuated, whereas the influence of the +1 and +2 positions are further decreased. All axes are scaled to the same size for each strain; shapes and amplitudes are therefore comparable (see Methods for more details).

https://doi.org/10.1371/journal.pcbi.1011799.g003

allow quantifying the general impact of chromatin remodeler deficiency. Here, we plot a positive contribution to the mean in magenta and a negative contribution in green. Lastly, the location-specific effect of the discriminator links spatial properties to the Pearson gene clusters, which describe likely similarity of nucleosome positioning mediated by chromatin remodelers. The impact of the discriminator is in the following given as a grey area around the mean, indicating more important regions when the margin is larger. The median profile of each cluster using the determined fCPs will be given again in orange and blue. This allows a comprehensive analysis of the impact of gene deletions or *rsc8* depletion with respect to the WT.

We can determine the importance of particular positions to separate the clusters as follows. The slope of the SVM indicates the contribution of each fPC to separate the clusters. For example, a 0° angle shows that the descriminator can be solely described by the second fPC; 45° suggest an equal contribution of both fPCs to separate the clusters; and 90° indicate that collective nucleosome phasing is exclusively dependent on the first fPC. Consequently, by linearly combining both fPCs together as implied by the slope (Eq 9), we can evaluate which positions along the profile are particularly important for the classification. Indeed, understanding the separating boundary is not straightforward. Although the median profiles for each profile can

WT is per definition equal 0.								
	WT	chd1	isw1	isw2	rsc8	isw1/chd1	isw2/chd1	chd1/rsc8
A	0.299	0.834	0.117	0.133	0.377	0.213	1.406	0.031
В	0.055	0.48	0.329	0.038	0.08	0.283	0.538	0.074
Mean µ	0.177	0.657	0.223	0.0855	0.2285	0.248	0.972	0.0525
s	0	2.6674	0.0409	0.3612	0.0366	0.2951	2.9842	1.4773
	isw1/isw2	isw1/rsc8	isw2/rsc8	isw1/isw2 chd1	isw1/chd1 rsc8	isw2/chd1 rsc8	isw1/isw2 rsc8	isw1/isw2 chd1/rsc8
A	1.452	0.347	0.153	0.112	0.216	0.057	0.466	0.066
В	1.074	0.072	0.567	0.295	0.207	0.068	0.245	0.174
Mean µ	1.263	0.2095	0.36	0.2035	0.2115	0.0625	0.3555	0.12
s	12.7873	0.0157	0.3315	0.0157	0.5420	4.8846	0.5909	0.1233

Table 1. SVM boundary slopes for both replicates. The first two rows give the boundary slope for replicate A and B, respectively. Mean μ is the mean slope for both. The s value represents our significance measurement defined in Eq.7. Noteworthy changes of the boundary slope are marked in green (bold), all others are red. The s-value in WT is per definition equal 0.

https://doi.org/10.1371/journal.pcbi.1011799.t001

differ substantially at some positions, this variance might be less important for separating and interpreting the clusters (e.g. the +2 nucleosome in WT conditions, Fig 3C). Reciprocally, whilst the median profiles for both groups can be very similar, the variance over all considered genes at this locus could be much larger and therefore play an important role for the classification (e.g. the +3 position in WT strains Fig 3C).

We identified 5 mutants—namely $chd1\Delta$, $isw2\Delta chd1\Delta$, rsc8-depleted $chd1\Delta$, $isw1\Delta isw2\Delta$, and rsc8-depleted $isw2\Delta chd1\Delta$ —that evoked notable changes considering the experimental variability between replicates (Figs 3–5). For a correct interpretation of the results, it is crucial to highlight that this does not imply that other mutants had no effect on the profile. Rather, this suggests that the considered mutation caused a gene-specific change of nucleosome phasing regulated by chromatin remodelers, which we assume is represented by the deviance of the stochastic process (i.e. the variance to describe the MNase-seq profiles). Other gene deletions can have other impacts that do not disrupt the gene-specific collective positioning. All measurements are given in Table 1.

Most single mutants had only a small or negligible effect on the collective nucleosome phasing along transcribed regions, with the exception of $chd1\Delta$ (Fig 3D–3F). Indeed, the boundary was notably tilted with respect to WT conditions (Fig 3D). This suggests that the functional composition of the MNase-seq signal changed. In fact, the amplitude of the second fPC decreases more quickly along the gene body in $chd1\Delta$ mutants, and the variance of the peak at the +2 position strongly diminished (Fig 3E). When interpreting the effect of the discriminating boundary, we observe that the +1 and +2 nucleosomes only exhibit a small importance for establishing the clusters, whereas the impact of the NDR and later nucleosomes increased (Fig 3F). Consequently, the +1 position remains largely unaffected. As Chd1 is responsible for nucleosome spacing along genes and is particularly involved in maintaining chromatin integrity during Pol II elongation, it is intuitive that the chd1-deletion influences phasing within the gene body. This outcome shows the clear effect of chromatin maintenance by Chd1 after the +2 nucleosome, whilst leaving the +1 position well preserved.

The double mutant *isw* $2\Delta chd1\Delta$ exhibited also a noteworthy shift of the separating boundary (Fig 4A), yet with different results to the *chd1*\Delta single mutant. The second fPC seemingly preserves its wave-like shape (Fig 4B). This indicates that nucleosome presence is less perturbed, and peaks are comparatively well positioned. Similar to the *chd1*\Delta single mutant, both of the fPCs strongly contribute to distinguish between the Pearson clusters. The discriminating function exhibits similar local effects as the *chd1*\Delta strain, but the positions after the +2 nucleosome clearly indicate an additional shift which contributes to the separation (Fig 4C).



Fig 4. The fPCs, their gene specific scores, and the discriminating boundary explain changing collective phasing in double mutants. The figure shows the cluster distribution with respect to ζ_{i}^{j} , the impact of the determined fPCs, and the location-specific impact of the separating boundary for all double mutants, in particular *isw2\Deltachd1\Delta* (i.e. panels A-C), $rsc8chd1\Delta$ (i.e. panels D-F), and $isw1\Delta isw2\Delta$ (i.e. panels G-I). The linear separation of the cluster distribution with respect to factors ζ_{i}^{j} indicate a notable gene-specific change for the three mutants in panels A, D, and G. The two clusters are given in orange and blue, and the SVM boundary is depicted by the black dashed line. Whilst *isw2\Deltachd1\Delta* and $isw1\Delta isw2\Delta$ require both fPCs to linearly separate the Pearson clusters, $rsc8chd1\Delta$ is almost exclusively dependent on the second fPC, which means this mutant decreased the slope tilt. This can be better understood when analysing the two fPCs and their effect on the mean ((B) for $isw2\Delta chd1\Delta$, (E) for $rsc8chd1\Delta$, and (H) for $isw1\Delta isw1\Delta$). The solid lines in magenta and green in these plots indicate a positive contribution of the fPC and a negative contribution, respectively, whereas the black dashed line depicts the mean. Grey arrows along the gene suggest general trends. Grey vertical bars suggest positions that remain largely unperturbed by the fPC. Grey arrows pointing to a single peak suggest remarkable properties. Interestingly, whilst the first fPC of the isw2\Deltachd1\Delta and isw1\Deltaisw2\Delta strains shows a similar transformation of the mean, the second fPC indicates a different behaviour, particularly with respect to the +2 nucleosome. As suggested by the fact that clusters in the *rsc8chd1* Δ mutant are exclusively dependent on the second fPC, the first fPC explains only the average profile amplitude and does not contain any information about collective phasing. The location-specific effect of the linear separator for each mutant is given in (C), (F), and (I). The grey areas indicate the importance of each position to determine the clusters, whose median profile is shown as a blue and orange dashed line. The mean is depicted in black. Although the impact on the grouping of the +1 and +2 position in *isw2\Deltachd1\Delta* conditions is similar to the *isw1\Deltaisw2\Delta* strain, the latter is seemingly particularly dependent on the +3 and +4 nucleosome. Positions thereafter become less important, which keep having a strong impact on clustering in $isw2\Delta chd1\Delta$. As expected $rsc8chd1\Delta$ is exclusively dependent on the second fPC. Interestingly, the entire profile seems to be influential for classifying genes, with the largest impact allocated to the first two nucleosomes. All axes are scaled to the same size for each strain; shapes and amplitudes are therefore comparable (see Methods for more details).

https://doi.org/10.1371/journal.pcbi.1011799.g004

Interestingly, *rsc8*-depleted *chd1* Δ significantly decreases the slope tilting (instead of accentuating it), therefore making coordinated phasing almost exclusively dependent on the second fPC (Fig 4D). This can be better understood when analysing their respective effects (Fig 4E). The first fPC solely explains average signal amplitude (which is not measured by the Pearson

correlation index) and hence contains almost no information about coordinated positioning. As expected, the local effect of the discriminating boundary follows the trend described by fPC2 (Fig 4F). The second fPC also indicates that the NDR before the +1 cannot be maintained (see arrow in Fig 4E and the grey area in NDR and +1 position in Fig 4F), which is in line with other studies [12, 40]. Remarkably, all nucleosome positions along the entire array seem to be important for the classification-particularly the first two-which is not the case for the other two double mutants. It should be noted that not all double mutants that include $chd1\Delta$ show a similarly notable tilting of the slope as the single mutant. This could possibly mean that these double mutants have opposing effects, although it is difficult to give a clear indication with the variation between only two replicates. We found an interesting behaviour for $isw1\Delta isw2\Delta$ (Fig 4G). The effect of the second fPC hints that the positioning of the +2 is strongly impacted, and following phasing becomes inharmonious (Fig 4E). The +1 is kept well positioned. The first fPC, on the other hand, resembles the first fPC of the *isw2\Deltachd1\Delta* mutant, with a minor difference at the +3 nucleosome (compare Fig 4E with Fig 4B). Indeed, when analysing the locationspecific properties of the separating function (Fig 4I), nucleosome profiles in the *isw1\Deltaisw2\Delta* strain seem to be clustered particularly with respect to a shift at the +3 and +4 position. This shift is apparently slightly corrected thereafter and becomes less important. Whilst seemingly similar, a shift in the *isw2\Deltachd1\Delta* strain after the +2 position remains important for the entire arrangement to determine the gene groups (compare Fig 4I with Fig 4C). This indicates that Chd1 and Isw1 contribute differently to nucleosome phasing in *isw2* Δ conditions, with the effect of Isw1 being possibly more confined. Taken together, these results show that double mutants can have varying and non-linear effects.

Among the triple and quadruple mutants, the only one that changed notably the clustering boundary is *isw2* Δ *chd1* Δ *rsc8* (Fig 5A). Once again, tilting is decreased. The effect of the fPCs and the separating boundary is almost identical to the *chd1* Δ *rsc8* mutant, suggesting that *isw2* Δ does not have a strong effect on the phenomenon (Fig 5B and 5C). However, it should be mentioned that the variability between the two replicates is considerably large, as the two clusters can be only neatly separated in replicate *B*, whereas replicate *A* exhibits a great overlap. Whilst the result in the latter replicate could suggest that more fPCs are necessary to interpret



Fig 5. The fPCs, their gene specific scores, and the discriminating boundary explain changing collective phasing in *isw2*\Delta*rsc8chd1*\Delta. The two clusters are given in orange and blue. The figure shows the fPC scores ζ of the *isw2*\Delta*chd1*\Delta*rsc8* mutant and their separating boundary (black dashed line, A). The slope decreases with respect to the WT, making the gene groups almost solely dependent on the second fPC. Both fPCs transform the mean in a similar way as the double mutant *rsc8chd1*\Delta (compare panel B with Fig 4E). The dashed black line as well as the solid lines in magenta and green indicate the mean, a positive contribution of the fPC, and a negative contribution, respectively. As expected, the separating boundary discriminate between the two clusters largely following the second fPC (C). The grey areas show the importance of each position to determine the clusters, whose median profile is shown as a blue and orange dashed line. The mean is depicted in black. All axes are scaled to the same size for each strain; shapes and amplitudes are therefore comparable.

https://doi.org/10.1371/journal.pcbi.1011799.g005

the gene groups, the results for replicate *B* indicate that sufficient information is preserved in the first two fPCs. More replicates would be needed to provide an answer. We also want to highlight that mutants with more than two gene deletions exhibited less clear nucleosome peaks, and a straightforward interpretation of the Pearson correlation with respect to the two discriminating trends (compare with cartoon in Fig 1D) could be difficult. The results for these strains should be taken with a pinch of salt.

Taken together, these outcomes show that remodeler mutants have varying effects on nucleosome positioning. Whilst most mutations do not notably alter the gene-specific nucleosome coordination with respect to the WT, we identified 5 mutants that exhibited a strong effect on phasing. Interestingly, most of them include $chd1\Delta$, which indicates an important role of Chd1 for local arrangement within the gene body. Using fPCA to visualise the Pearson clusters permits the clear and position-specific quantification of the induced impact among varying strains.

Pol II presence correlates with nucleosome organisation in $chd1\Delta$ mutants

In order to assess an interdependence of nucleosome organisation with other genomic properties, we compared the two Pearson clusters to Pol II levels, Sth1 occupancy, AT ratio over the entire gene, as well as upstream NDR length and orientation of the upstream NDR (i.e. tandem or divergent). We also included Mediator presence as a large protein complex with transient interactions predominantly at the NDR. All of these factors were clustered into two equallysized groups where possible. For example, Pol II presence along the gene was evenly separated into transcribed regions with high and low Pol II occupancy. Interdependence was measured by training a simple neural network with no hidden neurons using Hebbian learning [41]. Consequently, we assessed which nuclear groups (e.g. high or low Pol II presence) corresponded to which Pearson clusters. We want to stress that we did not aim to find a predictive model. Rather, this approach allowed us to measure a direct correlation between similarity of nucleosome phasing and other genomics properties. The initial k-mean clustering did not impose a constraint on the group size, and they could therefore differ in the number of genes they contained. To remove any prediction bias, we forced the clusters to be of the same size. Genes in the larger group with closest Pearson coefficient to all distributions in the smaller group changed the cluster. We found the analysis for WT conditions non-conclusive, and correlations varied between A and B replicate (S13 Fig top). Whilst A was slightly correlated with the AT sequence content (Fig 6A and 6B), this trend disappeared for B, and it might in both replicates rather correspond to the fPC orthogonal to the cluster boundary (S16 Fig). Overall, we were surprised that none of the investigated properties could indicate a clear interdependence with nucleosome phasing in WT (Fig 6A).

The correlation between positioning and other nuclear properties changed among the mutants (S13 Fig). The effect is particularly clear for $chd1\Delta$ (Fig 6C), as there is a strong interdependence between phasing and Pol II (Fig 6D), Mediator presence, and NDR size (S13 Fig). As aforementioned, Chd1 maintains, among others, chromatin integrity during Pol II elongation. The correlation is therefore in line with our previous conclusions and the function of Chd1. The established link between Pol II presence and nucleosome organisation remains conserved in all strains with a Chd1 gene deletion, except *isw1\Deltachd1Δrsc8*. This is similarly true for the correlation with Mediator occupancy and NDR length. There was also a slight correlation to Sth1 and AT ratio in cells containing *chd1*Δ, which were, despite being weak, still notably stronger than in WT. The results are in agreement with the effects of Chd1 on chromatin maintenance during gene expression.

Due to the Rsc8-mediated nucleosome organisation within transcribed regions, we were wondering whether there is an increased interdependence to NDR length or gene size. We can



Fig 6. Remodeler deletions have varying effects on the interdependence with other genomic properties. The orange bars in panels A, C, and E show the ratio of correct predictions, and blue bars are wrong guesses. As we distinguish between two clusters, the dashed black line at 0.5 indicates random guessing. The dashed grey line with black edging in panels B, D, and F display the linear boundary for the Pearson clusters. Panels A and B: WT conditions are seemingly correlated with the sequence composition. However, the results are different for the *B* replicate, and therefore non-conclusive. All possible correlations are surprisingly low. Panels C and D: $chd1\Delta$ mutants increase particularly their dependence on PoI II and other transcription-coupled properties, such as Mediator presence. Surprisingly, the mutant showed also an increased interdependence on NDR length. Panels E and F: despite the Rsc8-mediated gene limits, there is no correlation with coordinated nucleosome phasing and the size of transcribed regions or NDR length. Although Sth1 indicates a slightly increased interdependence, this cannot be confirmed when plotting the groups with respect to the Pearson clusters. This is in line with our hypothesis that positioning in different regions interfere, and therefore, nucleosome localisation become increasingly independent from region-specific factors.

https://doi.org/10.1371/journal.pcbi.1011799.g006

report that there is no correlation with NDR size in any *rsc8*-depleted strain (Fig 6E). This is in line with our hypothesis that Rsc8 decouples processes at different genes. However, *rsc8* mutant cells exhibited a slightly increased correlation with Sth1. By looking at the separation with respect to the Pearson cluster boundary, we find that there is no noticeable impact (Fig 6F). The results indicate that any correlation with region-specific properties is lost, which is likely due to the interference of nucleosome positioning of various regions.

isw1 Δ single mutant did exhibit only a slightly increased correlation with Pol II, Sth1 and Mediator presence as well as AT ratio. *isw2* Δ might as well show a weak correlation with Pol II occupation. Each of the replicates of their double mutant indicates different correlations, and it is therefore difficult to tell whether transcription-related factors influence nucleosome phasing in the *isw1* Δ *isw2* Δ strain. However, none of them indicate any strong interaction, suggesting that—on a global scale—these effects might be negligible in comparison to the WT (S13 Fig).

Interestingly, the *rsc*8-depleted *isw* 2Δ indicated a strong correlation with Sth1 and Mediator presence as well as NDR length. The effect was observable in almost all strains that contained

the double mutation with the exception of the quadruple mutant (S13 Fig). Taken together, this could indicate an impact along the gene body and the promoter region in strains that contain *rsc8*-depleted *isw2* Δ .

Surprisingly, combining two factors together (e.g. Pol II presence and AT ratio) to predict Pearson clustering did not increase accuracy. Instead, one factor dominated the correlation measurement, e.g. Pol II presence for $chd1\Delta$ strains. This could possibly suggest that—despite several factors showing increased interdependence—they can be reduced to a main influencing factor (which is not necessarily one of the tested properties).

Taken together, the results indicate a strong interdependence between local genomic properties—such as presence of large protein complexes or NDR length—and strains containing $chd1\Delta$. This supports our hypothesis of Chd1 being responsible for local nucleosome coordination.

Discussion

In this work, we analysed the collective positioning of nucleosome arrangement within the gene body in WT and chromatin remodeler-deficient strains by combining clustering of Pearson coefficients with fPCA, the latter being an analysis framework for functional data. Although fPCA is well established in the assessment of time series, it has not been previously used to understand location-specific nucleosome profiles on a global scale. As we argue that the Pearson index measures similarity of nucleosome arrangement between genes, we interpreted the effect of chromatin remodelers on the positioning by visualising the distribution of two established significant Pearson clusters using fPCA. Indeed, we can show that the sets of genes for all mutants can be sensibly separated by the two fPCs that explain most variance in the data, and more fPCs are not necessary to describe the clusters. This allowed the quantification of the effect on coordinated phasing. The significant Pearson groups were compared with other nuclear properties—such as Pol II presence and NDR maintenance—and sequencedependent characteristics. None of the commonly supposed influencing factors can easily explain coordinated nucleosome positioning in WT conditions. However, correlation between tested properties and phasing increases with some gene mutations. The analysis reveals the impact of different gene deletions of chromatin remodelers on nucleosome arrangement within the gene body. It shows Rsc8-defined boundaries for nucleosome positioning along the gene, suggesting a global impact over the entire array for each gene. On the other hand, the results for most strains that contained a Chd1 deletion indicated gene-specific local effects, which correlate largely with Pol II presence. In the following, we critically discuss the results and their significance.

We applied a pairwise Pearson cross-correlation index to measure profile similarity between genes. The linear correlation measurements evaluate the overall trend of the signal (i.e. increasing or decreasing distributions at similar positions), and it does not take signal scaling into account. Therefore, it assesses whether genes are apt to contain nucleosomes at similar positions. Indeed, similar nucleosome phasing could indicate similar but gene-specific chromatin remodeler dynamics, which justifies the rationale for measuring classical linear correlation. It also follows previous analyses using comparable measurements [33–35].

We classified genes according to their Pearson similarity by applying a *k*-mean clustering approach. *k*-mean was repeated over several random initialisations, therefore removing any prior bias. We used a silhouette criterion value to determine the best number of clusters, which was shown to be 2. It should be mentioned though that the cluster distribution according to the fPCA did not show a clear separation of the data points themselves (i.e. there were no distinct data accumulations). Thus, this clustering is imposed by our assumptions using the Pearson index. Nonetheless, we argue that they reveal important information about

nucleosome phasing linked to chromatin remodelers when compared with mutant strains. Grouping of nucleosome profiles has been previously performed using different algorithms [42], and it is necessary to show that a separation along the first two fPCs is similarly possible when using other clustering approaches. We therefore compared the results of the Pearson kmean clusters with a WARD algorithm using the Euclidean distance metric. Once again, profiles can be best separated into two clusters (S14(A) Fig). Moreover, the gene groups tend to separate along the first fPCs, despite the fact that the boundary is less neat (S14(B) Fig). It should be mentioned though that clustering with an Euclidean distance measures different properties of the profiles, and it is therefore expected that the separation border changes. Indeed, the separation occurs now predominantly with respect to fPC1 (S14(B) Fig), which indicates a strong influence of the signal scaling that was ignored by the Pearson correlation. Therefore, the clustering does not express similarity of the overall nucleosome positioning along the entire array. We want to emphasise that we were particularly interested in how the entire nucleosome array tends to behave *as a whole*, which should be sensibly measured by using the Pearson correlation. We decided to ignore the scaling of the sequencing amplitude, as it conveys how many cells contain a single well-positioned nucleosome instead of how nucleosomes tend to behave with respect to most cells. Nonetheless, the found gene sets based on a fundamentally different clustering method tend to separate into different areas when using the first two fPCs. We conclude that the observed separation using *k*-mean clustering and the Pearson correlation as a distance metric captures genuine biological properties.

The determined gene groups were as different as possible using similarity measured by Pearson correlation. The validation using the silhouette criterion together with the KS significance test over the inter-cluster correlation between the found *k*-mean clusters and random grouping showed their significance. This shows that the data could not be better categorised using linear correlation. Significance tests are often prone to attributing a high importance to small differences when sample sizes are large. Therefore, we randomly sub-sampled 500 inter-cluster correlation indices for the determined and random grouping, respectively. It should be mentioned that in a few single instances, the p-value was larger than 5%, as we randomly selected 500 similarly low values for both inter-cluster correlations (i.e. for the *k*-mean and the random clusters). We accounted for these fluctuations by averaging the p-value over 500 random clusters from which each 500 random sub-samples were selected. Using this approach, we found that the *k*-mean clusters were significant for large genes of all mutants (average p-value <5%).

As the Pearson correlation index only indicates average similarity over the entire nucleosome array, we aimed to compare the clusters to the data itself in order to interpret their differences. Dimensionality reductions are often used to visualise clusters, such as for single-cell sequencing analyses [43]. Common approaches include PCA, uniform manifold approximation and projection (UMAP) [44], and t-distributed stochastic neighborhood embedding (t-SNE) [45]. Whilst the latter two are non-linear dimensionality reductions, PCA and fPCA find a linear decomposition of the data into the axes (or functions) that explain most variance. It is challenging to retrieve the exact meaning of the discriminating boundary using non-linear approaches. Consequently, understanding the location-dependent differences in the profile between two clusters and interpreting their separating function is more straightforward for PCA or fPCA than for UMAP and t-SNE. Although PCA and fPCA are very similar, PCA assumes that every position in the MNase-seq data is independent, whereas fPCA conjectures that they were produced by a single stochastic process along the spatial axis. Therefore, positions are dependent on each other. This is in line with the barrier model for establishing nucleosome phasing, which makes fPCA preferrable over PCA. Moreover, as we treat each nucleosome profile as one entity by using the Pearson correlation, the independence

assumption would violate the fundamental understanding in our analysis. Nonetheless, when comparing PCA and fPCA, we showed that the two clusters can be similarly separated (S15 Fig), although the two principal axes are slightly differently shaped due to the missing constraint of the spatial dependence.

FPCA assumes a stochastic process with a mean behaviour over the entire data set, and it characterises each data point with respect to their deviance from that mean (see Methods). The results therefore depend on the entire considered data set. Indeed, we find different results when including all genes or exclusively transcribed regions >1000 bp. However, these differences are not strong. Moreover, any possible bias was excluded by removing genes smaller than 1000 bp from a subsequent analysis. Due to the abundant and well-positioned nature of nucleosomes within the gene body in *Saccharomyces cerevisiae*, we find it justified to presume an average nucleosome distribution describing their wave-like profile. Nonetheless, we argue that the variance between genes contains important information about nucleosome phasing imposed by chromatin remodelers, which we roughly categorised into groups. We found that the two Pearson correlation clusters could be neatly separated by the fPC scores ζ_{i}^{i} , $j \in \{1, 2\}$. This indicates firstly that the Pearson index measures a trend that is explained by the largest variance in the data; and secondly, the two fPCs that describe most variance are sufficient to interpret the clusters.

Whilst linear-correlation measurements are limited to quantifying the average similarity, a combination with fPCA allows characterising location-specific differences and in which way gene deletions affect phasing from an average. Evaluating the effect of the linear boundary along positions within the gene body revealed detailed differences in the nucleosome profile that are important for establishing the groups. As our approach is largely dependent on general signal processing methods, we can repeatedly apply the same framework for all mutants and compare there results. Therefore, the combination of linear correlation with fPCA extends previous ways of analysing nucleosome distributions using only correlation measurements that average over the entire profile [33–35].

The analysis can clearly distinguish between mutant-specific effects on phasing. All mutants preserved the information of coordinated nucleosome arrangement in their first two fPCs, and the Pearson clusters could be separated by a neat line. Consequently, none of the chromatin remodeler gene deletions caused random positioning. Some mutants, however, showed an increased overlap between the two groups, which indicates increased independence between individual nucleosome locations, and positioning might be more random. Including more fPCs could help further separating the clusters. In all of those cases however, one of the two replicates always permitted a clearer separation by using only the first two fPCs. Considering the experimental variability in the data, it is not possible to draw direct conclusions without further replicates. In order to simplify the comparison between mutants, we restrained from including more fPCs.

Most strains did not alter notably their gene-specific collective arrangement (i.e. the slope), and a linear separation of the Pearson clusters using the deviance from the mean did not change with respect to WT strains. Although they can nevertheless have an impact on the mean itself, coordination along the genes remains preserved in a similar way, at least as measured by the Pearson correlation index. Due to the focus of the study on coordinated nucleo-some positioning along transcribed regions, we did not consider them as having notably changed their coordinated phasing.

The length of an average yeast gene is approximately 1500 bp, which corresponds to 6–7 nucleosomes. To analyse the effect of nucleosome remodeler deletions or their depletion on their arrangement, we examined a stretch of 1200 bp per gene, including 200 bp upstream of

the +1 dyad. By considering an array of 6–7 nucleosomes at the same time, we were able to assess their long-range impact on the gene body. During our analysis, we differentiated between small and large genes (i.e. coding regions shorter than the considered 1000 bp after the +1 dyad versus genes \geq 1000 bp). This revealed the effect of Rsc8 on limiting nucleosome positioning to the gene body. It should be emphasised that we removed smaller genes from the subsequent analysis, as it would be impossible to compare nucleosome arrangement further down into the gene body. We also surmised that the influence of remodeler deletions on later nucleosomes is more difficult to observe in smaller genes, since they could be located very right at the TTS which could have an additional impact. Therefore, we based our analysis and conclusions—particularly with respect to Chd1—only on genes larger than 1000 bp. It should be noted that the results may not be easily transferable to small genes.

Gene mutations of chromatin remodelers have been analysed previously in detail, including their influence on phasing [12–14, 18], NDR maintenance [46], and gene transcription [13]. RSC is the only essential chromatin remodeler complex in *Saccharomyces cerevisiae* [47], and it has been particularly associated with positioning of the +1 and -1 nucleosomes [12, 46, 48]. This mechanism has been proposed to be conserved among various yeast species [11]. It has also been reported that RSC regulates expression of Pol II and Pol III-transcribed genes [13, 49, 50]. Moreover, it has been found to impact Pol II elongation and termination [12]. All of these results imply that RSC is to some extent involved in limiting the transcribed region. However, this has been predominantly quantified with respect to changes at the core promoter. To our knowledge, a potential role for Rsc8 to decouple nucleosome phasing in independent genes has not been suggested. The presented functional analysis of MNase-seq profiles in rsc8-depleted strains clearly indicates a coordinated nucleosome arrangement that exceeds the limits of transcribed areas. This is further supported by our finding that correlation with other nuclear and sequence-dependent factors decreases. Furthermore, mutants that were rsc8 depleted decreased notably the boundary slope between the two clusters, indicating that coordinated positioning becomes increasingly independent of other functional components. The strictly limited and Rsc8-mediated phasing barrier could have further implications for other processes—such as transcription—as nucleosome placing in one gene influences its neighbouring regions. The notion of gene-interfering positioning has been also proposed by [14]. The study shows that RSC could act as a bidirectional barrier, influencing upstream and downstream regions. Interestingly, they found that interference also plays a crucial role in WT strains, and that the same phenomenon remains preserved in rsc8-depleted cells. However, our fPCA reveals that the limiting role of the RSC remodeler complex is crucial in WT conditions, and that this behaviour is significantly altered when Rsc8 is depleted. Taking this into account, Rsc8 should fulfill the role of disentangling gene-related processes in WT strains, and it therefore allows for a flexible and uncorrelated transcriptional program. Indeed, rsc8depleted cells exhibit significantly altered Pol II profiles [10, 12], which is in accordance with our hypothesis. We propose that the RSC chromatin remodeler globally disentangles nucleosome phasing, and it therefore plays a substantial role in long-range positioning.

Interestingly, our results indicate that positioning limited to the gene body can be re-established in *rsc8*-depleted *chd1* Δ mutants. We hypothesise that they have antagonistic effects in establishing gene size-dependent barriers for nucleosome arrangement. Indeed, it was reported that Rsc8 and Chd1 have opposing effects for Pol II termination. *rsc8*-depleted cells exhibit inhibition of Pol II dissociation at the TTS, whereas the double mutant *isw1* Δ *chd1* Δ increases release frequency, with seemingly *chd1* Δ dominating this effect [12]. The authors propose that this is related to the close packaging of nucleosomes at the TTS. Our outcomes suggest that they might have antagonistic effects in chromatin organisation that differs between transcribed and non-transcribed regions.

We found that $chd1\Delta$ mutants had a strong impact on coordinated positioning within the gene body. Indeed, Chd1 has been, among others, characterised with respect to its role in maintaining chromatin integrity during Pol II transcription [16, 51, 52], and it associates to both promoters and transcribed regions [53]. This is in line with our finding that correlation with Pol II presence and occupancy of Mediator increases in Chd1-deficient strains. With the exception of *isw1\Deltaisw2\Delta*, all other noteworthy changes included deletion of *chd*, further emphasising its role for chromatin organisation within the gene. However, not all chd1Δ-containing mutants exhibit a notable effect. This can have various reasons, including experimental variability. However, particularly the mutant $chd1\Delta isw1\Delta isw2\Delta$ could indicate an interacting behaviour of the remodelers. Indeed, Chd1 has been reported to cooperate [16] as well as antagonise Isw1 [18], and therefore could have different effects depending on the context. With this being said, the behaviour of the triple mutant $isw1\Delta isw2\Delta chd1\Delta$ is particularly interesting, as $chd1\Delta$ and $isw1\Delta isw2\Delta$ each individually affect coordinated phasing, but not their triple mutant. This could suggest an antagonistic behaviour on nucleosome coordination. As Chd1 is highly conserved in all eukaryotes [54], this result could have consequences beyond Saccharomyces cerevisiae.

It is worth noting that the Isw1 subunit is part of the two complexes Isw1a and Isw1b, and consequently its deletion affects them both. Previous studies [55] have shown that these complexes have different functions and enrichment profiles. The Ioc3 subunit of the Isw1a complex is enriched at +1 nucleosome, while the (Isw1b) is enriched on +2, +3, and +4 nucleosomes. In addition, these two complexes can act on different sets of genes and promote sliding in different directions. However, as we did not analyse the effect of specific subunits of Isw1a or Isw1b complexes in this study, we cannot distinguish between the two complexes. Therefore, it is possible that the effects observed from the +2 nucleosome onward may not be due to one specific complex.

Analysing the MNase-seq data using fPCA allowed us to obtain a different view on the functionality of various remodelers to maintain chromatin organisation. We propose the following mechanism (Fig 7). The RSC remodeler complex is essential for allowing independent



Fig 7. Chromatin remodelers maintain nucleosome organisation on a local and far-reaching scale. Top: RSC (green ellipse) establishes independent nucleosome phasing on each gene (two vertical dashed lines) by maintaining the NDR through positioning the +1 (cornered arrow) and -1 nucleosome. The ATP-dependent positioning is symbolised by black arrows pointing away from RSC. The local remodeling effect of Chd1 (blue ellipse) allows chromatin arrangement independent of Pol II transcription (yellow ellipse). Bottom: in *rsc8* strains, the NDR cannot be maintained anymore, and phasing in and outside a gene interfere with each other (single dashed line). We propose that this should equally lead to an increased interdependence of other nuclear processes such as transcription. If *chd1* is deleted, nucleosome arrangement is more sensitive to the presence of other large complexes, such as Pol II. During transcription, Pol II is affecting the local positioning (black arrows from Pol II).

https://doi.org/10.1371/journal.pcbi.1011799.g007

phasing in each single gene. It plays therefore a pivotal role in maintaining the barrier with respect to which nucleosome positioning is coordinated. This permits the decoupling of gene-specific processes such as transcription. Depletion of Rsc8 leads to the interference of different genomic regions, which therefore alters sequence accessibility on a global scale. Indeed, it has been reported that gene expression is dramatically changed in *rsc8* mutants [10, 50]. Chd1, on the other hand, maintains chromatin integrity during transcription [16, 51, 52], and it influences nucleosome phasing locally to permit Pol II-mediated expression. *chd1* Δ strains make positioning dependent on Pol II presence. Consequently, whilst RSC plays a global role, Chd1 is important for local nucleosome organisation.

Methods

Data treatment

MNase sequencing reads were taken from [18] and [12] (GEO accession numbers GSE69400 and GSE73428, respectively) and treated as in our previous study [56]. To be precise, reads from Fastq files were trimmed with trim_galore (v0.6.5) [57] and cutadapt (v3.1) [58]. Subsequently, they were mapped on the *Saccharomyces cerevisiae* genome (University of California at Santa Cruz [UCSC] version sacCer3) using bowtie2 (v2.3.4.3) [59]. Files were converted with samtools (v1.9) [60] and deeptools (v3.5.0) [61]. Read counts were normalised in Reads Per Million (RPM) of mapped reads. We used the option --MNase of bamCoverage so that only the mononucleosome fragments were kept. This means that fragments shorter than 130 bp and longer than 200 bp were removed from analysis. Mediator and Sth1 ChIP-seq were taken from [56] (ArrayExpress accession number E-MTAB-12198). We used Pol II ChIP-seq from our previous study [62].

Following [12, 18], we retrieved positioning profiles along the coding regions 200 bp before and 1000 bp after the +1 nucleosome. Genes on the Crick strand were inverted. Consequently, all data is calibrated such that the +1 position is at 200 bp. The profile of genes for which the +1 position is known were considered as in [18].

Measuring profile correlation and clustering

The pairwise Pearson correlation of MNase-seq distributions for each gene was determined using equation

$$r_{xy} = \frac{\sum_{i=1}^{n} (x_i - \bar{x})(y_i - \bar{y})}{\sqrt{\sum_{i=1}^{n} (x_i - \bar{x})^2} \sqrt{\sum_{i=1}^{n} (y_i - \bar{y})^2}}.$$
(1)

Here, *x* and *y* denote two genes, \bar{x} and \bar{y} symbolise their respective average MNase-seq value along the coding region, and *n* = 1200 is the length of the considered region. Eq 1 ranges between -1 and 1, and indicates whether the two gene profiles tend to change into the same (positive Pearson correlation) or opposite directions (negative Pearson indices).

Genes were divided using the *k*-mean clustering implementation in MATLAB with the correlation distance metric. To define the optimal number of *k*-mean clusters, we used the silhouette criterion measurement [63, 64]. For all analysed strains, the highest silhouette value occurs at 2 groups, suggesting that in order to divide the profiles into classes with respect to their Pearson indices, the optimal number of clusters is 2 (Fig 1A). Therefore data were grouped in two clusters (Fig 1B and 1C).

Cluster significance was validated by comparing the inter-cluster correlation (i.e. each pairwise Pearson index between a profile in cluster 1 and a profile in cluster 2) between the

determined and random clusters. It is expected to be minimal for the gene groups found by the *k*-mean clustering. Therefore, we applied the KS test implemented in scipy with the hypothesis that the inter-cluster correlation of the found gene sets is significantly lower than for random groups. It should be noted the number of all pairwise comparisons can be quite large. It is a known problem that small differences are marked as being significant by the KS significance test when sample sizes are too large. Therefore, we sub-sampled randomly 500 pairwise inter-cluster correlations for the determined and the random clusters, respectively. It should be mentioned that this sometimes led to very similar sub-samples, and the p-value was for some few runs fairly high. We therefore averaged the p-value over for the KS test over all 500 random clusters and used the conventional p-vale threshold of 0.05 (i.e. 5%) to reject the null hypothesis. To be more precise, we required that the average probability of finding random clusters that have a similarly low inter-cluster correlation was lower than 5%. This was the case for all tested mutants.

Functional principal component analysis

Functional clustering in a Hilbert space *H* can be achieved by fPCA. It applies—similar to PCA in Euclidean space—a functional dimensionality reduction in *H* to investigate the dominant mode in functional data. Instead of relying on values in discrete dimensions, fPCA uses a given number of basis functions (e.g. B-splines or Wavelet) to create the eigenfunction basis that accounts for most functional variation. Despite the fact that MNase-seq data is stored in a discrete array (i.e. one value per bp), we can nevertheless find a functional approximation over a range using a given choice of basis functions. It should be noted that this implicitly smooths out high frequencies in the signal. We presume that nearby values in MNase-seq data possess a strong interdependence, therefore justifying a smoothed and continuous functional representation of the high dimensional data. In this study, we apply B-splines as a basis to represent the nucleosome array (Fig 8). We use the Python library scikit-fda to determine the fPCs and the corresponding weights explaining the distribution [65]. Here, we describe briefly the underlying principles of the method.

FPCA presumes that the functional data represents a stochastic process X(t) with expected value $\mu(t) = E[X(t)]$ and orthonormal eigenfunctions $\phi^i(t)$, i = 1, 2, ... Intuitively, $\phi^i(t)$ describes the most variation in X orthogonal to all ϕ^i , j < i. This allows the iterative determination of the eigenfunctions in the functional data. It should be emphasised that in this study the process is defined in space rather than describing temporal data. We follow nevertheless the convention by denoting the independent variable as t. By using the Kosambi–Karhunen–Loève theorem, any stochastic process can be represented as an infinite linear combination



Fig 8. Representing the MNase data array as a composition of B-spline base functions in WT conditions. Left shows the raw data, each colour depicting one profile over a gene. Center gives the smoothed profiles after representing the data as B-splines. Right displays the average profile using the functional composition.

https://doi.org/10.1371/journal.pcbi.1011799.g008

 $\phi^{i}(t)$. Consequently, we can describe the stochasticity in X(t) via

$$X(t) - \mu(t) = \sum_{k} \zeta^{k} \phi^{k}(t).$$
⁽²⁾

 ζ^k is the autocovariance operator

$$\zeta^{k} = \int (X(t) - \mu(t))\phi^{k}(t)dt.$$
(3)

To provide some intuition, it is presumed that the entire data set can be explained via an average behaviour $\mu(t)$. Variability to $\mu(t)$ for each gene is expressed by $\phi^k(t)$ together with a factor ζ^k . ζ^k can be loosely compared to a normal correlation measurement, i.e. ζ^k increases when $\phi^k(t)$ and $\int (X(t) - \mu(t))$ follow the same trend. If they describe opposing behaviours—for example $\phi^k(t)$ decreases when $\int (X(t) - \mu(t))$ increases— ζ^k becomes negative.

It is commonly justified to approximate Eq 2 as a finite sum

$$X(t) \approx X_n(t) = \mu(t) + \sum_{k=1}^{n} \zeta^k \phi^k(t).$$
(4)

It should be noted that $\phi^i(t)$, i = 1, 2, ... is a basis of the functional space in *H*.

This understanding of the underlying process permits the application of fPCA. A smoothed representation with the basis functions (e.g. B-splines) fulfilling Eq.4 can be obtained using L^2 regularisation. To reduce the dimensionality to *K*, we keep only the first *K* components (i.e. $\phi^i(t)$) that represent the dominant mode of variation in *X* by setting the first component to

$$\phi^{1} = \arg\max_{\|\phi\|=1} \left\{ \operatorname{Var}(\int_{\mathcal{T}} (X(t) - \mu(t))\phi(t)dt) \right\},\tag{5}$$

and the following K - 1 components to

$$\phi^{k} = \arg\max_{\|\phi\|=1,\langle\phi,\phi^{j}\rangle=0} \max_{\text{for } j=1,\dots,k-1} \left\{ \operatorname{Var}(\int_{\mathcal{T}} (X(t) - \mu(t))\phi(t)dt) \right\}.$$
(6)

 $\|\phi\|$ is the square norm, i.e. $\|\phi\| = \sqrt{(\int \phi(t)^2)}$. It should be emphasised that ϕ^k can differ by a factor of -1 due to the square norm, and consequently, the operator ζ^k (Eq 3) can be either positiveor negative depending on ϕ^k . This means that the slope of the cluster-dividing boundary can be pointing upwards or downwards and still describe the same functional composition.

We exemplified the impact of the first two fPCs to analyse the consequences on nucleosome phasing in chromatin remodeler-deficient cells (see for example Figs 1, 2 and 4). It should be noted that the fPCs were amplified to highlight their functional contribution. We set the scaling factor to $\zeta^1 = \zeta^2 = 20$ in all figures that demonstrate their effects (i.e. magenta shows the effect of the fPC multiplied by 20 and added to the mean, and green depicts the fPC multiplied by 20 and subtracted from the mean). The determined factors were predominantly distributed in $\zeta^{1, 2} \in [-20, 20]$ for all strains and replicates, and most of them were in fact much lower. Therefore, we limited the scaling of the axes for ζ^1 and ζ^2 to [-20, 20] for all plots that show the cluster distribution with respect to the factors. Therefore, all figures and axes were directly comparable. The few outliers that were outside this range were incorporated into the analysis despite of being not shown in those plots. In most cases, only 1 or 2 genes were outside these ranges. However, large genes for *isw1* replicate A and *isw2 Arsc8chd1* replicate B had 71 and

72 outliers, respectively. Nonetheless, this value amounts to less than 2%/ of all considered protein-coding profiles for these setups.

In order to validate that nucleosome arrangements tend to separate along the first two fPCs when using a different clustering method, we repeated grouping using WARD with an Euclidean distance metric on the profiles. Clustering was implemented using the AgglomerativeClustering class of the scikit-learn Python package. In contrast to the Pearson correlation index, the Euclidean distance does not have an upper bound, and clustering was more sensitive to outliers. We removed 29 genes who had an absolute fPC score larger than 20 for either of the two fPCs. Despite the fact that the boundary is less neat than when using the *k*-mean clustering, our results suggest that the WARD gene groups tend to separate along the first two fPCs as well (S14 Fig).

Quantifying the cluster boundary

Long genes were linearly separable with respect to the Pearson coefficient clusters in all WT and mutant conditions. The boundary was determined using a linear SVM. We ignored the prediction error and the intercept of the linear boundary, and instead considered only the slope differences between the two replicates. As aforementioned, the sign of the slope *m* does not matter, and we consider therefore only |m|. To quantify the variability in the two replicates, we introduce the following measurement

$$s(i) = \frac{(\bar{m}_i - \bar{m}_{WT})^2}{(|m_i^A| - |m_i^B|)(|m_{WT}^A| - |m_{WT}^B|)}.$$
(7)

 \overline{m} denotes the average over the absolute slopes of both replicates. We defined a change as notable when s(i)>1, which implies that the mean variability between WT and mutant is larger than the variability within the replicates, i.e.

$$\left(\bar{m}_{i}-\bar{m}_{WT}\right)^{2} > \left(|m_{i}^{A}|-|m_{i}^{B}|\right)\left(|m_{WT}^{A}|-|m_{WT}^{B}|\right).$$
(8)

As we consider only two replicates, we restrain from using the word *significant* as much as possible and use *noteworthy* or *notable* instead.

The slope of the boundary *m* indicates the contribution of each fPC to describe the discriminator between the clusters. As *m* shows the change of ζ^2 over one unit of ζ^1 , we can determine the separating boundary by

$$\phi' = \frac{m\phi^1 + \phi^2}{m+1}.$$
(9)

The impact of ϕ' can be visualised by multiplying a scaling factor which is followed by addition to and subtraction from the mean. In this study, we used a factor of $\zeta' = 5$ to create the grey bands in the plots that show the effect of the separating function.

Measuring interdependence between nucleosome phasing and other nuclear properties

In order to analyse interdependence of nucleosome positioning with other nuclear properties, we divided all factors into two equally sized cluster using the median wherever possible. For example, the half with the smaller NDRs was assigned to group -1, whereas the larger half was group 1. This split was performed after filtering for the size (i.e. large or small genes). The analysis aimed to find a correlation between nuclear factor group and Pearson cluster. To remove

any bias with respect to the group size, we forced both Pearson clusters to contain the same number of genes.

We used a simple feedforward network with no hidden neurons and a single output neuron whose activation indicated the predicted Pearson cluster. The number of input neurons varied between 1 and 2, depending on whether we considered a multivariate interdependence. The group of the nuclear factor (i.e. -1 or 1) was set as input neuron activation. This was weighted and summed together with all other input values. The activation function of the output was a modified sign function, which returned 0 when negative and 1 when positive. Therefore, if the weighted sum over the input was lower than or equal to 0, the output would be 0, and 1 otherwise.

Weights were trained using a Hebbian-like learning method [41]. In order to avoid any confusion, we name Pearson cluster 0 and nuclear factor group -1 *low* cluster, whereas we define group 1 in both cases to be the *high* cluster. The weight was defined to be the average number of genes where the nuclear factor group and Pearson coefficient cluster where both *low* or both *high*; minus the average number where one of them was *low* whilst the other *high*. The implementation as a neural network allowed the straightforward extension to compare interdependence with several factors at the same time using the same method.

Supporting information

S1 Fig. Cluster significance test for all genes in WT replicate A. We measure the inter-cluster correlation for the found clusters and 500 random clusters. It is expected that the found gene groups have a significantly lower inter-cluster correlation than a random separation (i.e. the found clusters are most dissimilar based on the Pearson correlation). A KS test over 500 randomly sampled inter-cluster correlation indices for the *k*-mean and random grouping, respectively, proved their significance. Magenta shows the distribution for the found clusters, orange displays one example for random clustering. (A) The distribution over the absolute pairwise inter-cluster correlations is significantly lower for the gene groups determined by the *k*-mean clustering than for the random grouping. (B) The cumulative distribution function for the found clusters raises much more quickly. A KS test verified that this trend is indeed significant (average p-value over 500 repetitions 0.0009). (TIFF)

S2 Fig. The WT fPC scores ζ coloured with respect to the Pearson clustering using all genes. Blue and orange indicate each one group, the dashed line symbolises the best linear separation using a SVM. The x-axis represents the score of the first fPC ζ^1 , the y-axis gives the score for the second fPC ζ^2 . All axes are scaled to the same size; shapes are therefore comparable. (A) and (B) show all genes for replicate A and B. (C) and (D) display the fPC scores after filtering for large genes (>1000 bp) for replicates A and B. (E) and (F) show small genes (($\leq 1000 \text{ bp}$) for replicate A and B. We removed the separating boundary because it did not reasonably divide the clusters. Nevertheless, we kept the estimated linear function in the legend to allow a comparison with other boundaries. Of particular note is the bias, which can be even order of magnitudes different from large-gene clusters. (G) and (H) display the fPC scores after filtering for very large genes (>3000 bp) for replicates A and B. (TIFF)

S3 Fig. First 10 fPCs for WT A all genes. FPCs are ordered with respect to how much variance they explain (i.e. fPC1 explains the most whereas fPC10 explains the least). The mean is given as a black dashed line, a positive contribution is shown in magenta, whereas a negative contribution is displayed in green. The first two fPCs are the ones that were presented in Fig 1

(F) (21.3% and 11.5% explained variance, respectively). The fPCs that follow after the major two ones become increasingly complex, and it is difficult to quantify their effect in a straightforward measurement. With the exception of fPC4 (7.8% explained variance), the plots suggest that the effects of the fPCs that were not included would not have been captured by the linear correlation index, as they describe changes specific to a single nucleosome (e.g. fPC6) or complex changes (e.g. fPC7). We want to remind that the variance captured by the fPCs could either moderately occur along the majority of genes; or alternatively, there is a strong effect on a small subset of profiles. We interpret the results as follows. FPC1 and fPC2 show global trends how nucleosome arrangements change along most protein-coding regions. This is emphasised by the fact that they capture the most deviance from the mean and that their effects are not specific to particular nucleosome positions. Many other fPCs (e.g fPCs 5, 6, and 7, 7.5%, 7%, and 6.1% explained variance, respectively) include strong position-specific effects. Despite the large impact on the amplitude at precise positions (e.g. +3 or +4), the explained variance by these fPCs is lower. We presume that this indicates a strong variance at these positions in a small subset of genes. Thus, they show a much lower variance. We focus on the first two fPCs because they can separate the Pearson clusters, which in turn indicates that they capture global trends since they were determined over all genes. (TIFF)

S4 Fig. Heatmaps for small-gene nucleosome profiles reveal antagonistic roles for Rsc8 and Chd1 to establish phasing boundaries. Cluster 1 and 2 for all genes in WT conditions were plotted only including small genes on the left. Indeed, correct positioning is either completely disrupted (Cluster 1), or clear phasing is lost after +3 or +4 position and individual peaks do not stand out thereafter (Cluster 2). However, both Pearson clusters for *rsc8*-depleted cells (centre) show clear phasing probabilities, despite all genes being smaller than the considered 1000 bp after the +1. The double mutant *chd1*\Delta*rsc8* seems to re-establish the gene boundaries for nucleosome phasing, as positioning is either disrupted (Cluster 2, compare with Cluster 1 in WT) or does not exhibit clearly distinguishable peaks after the +3 or +4 nucleosome (Cluster 1, compare with Cluster 2 in WT). Defining a group as being 1 or 2 was arbitrary and has no significance. Copper values show large MNase-seq signal values, whereas dark segments indicate a low amplitude. Values in between are uniformly scaled. (TIFF)

S5 Fig. The large-gene fPC effect in WT. Despite fact that the functions differ in the A and B replicate ((A) and (B)), they both describe the same properties as when considering all genes (Fig 1(F)). To be precise, the first fPC describes seemingly position-dependent scaling (grey vertical bars), and the second explains coordinated phasing (grey arrows). The mean is displayed as a black dashed line, whereas a positive and a negative functional contribution are given in magenta and green, respectively. (TIFF)

S6 Fig. The Pearson coefficient clusters for exclusively small genes correspond to the gene size. When we repeated the Pearson coefficient clustering considering exclusively small genes, we can linearly separate again the two groups (orange and blue). However, this is predominantly explained by the size of the gene (short pink, long green). This in line with the hypothesis that coordinated nucleosome phasing along the transcribed region is strictly limited within the gene body. The phase separating line was determined on the Pearson clusters (dashed black line) using an SVM. The same separating boundary was also plotted in right plot showing grouping with respect to the size. We plotted the original SVM boundary from the Pearson clusters with a dashed grey line to indicate that it was not determined using gene size. (A) and (B) give the Pearson clusters for replicate A and B. (C) and (D) show the size dependence of replicate A and B.

(TIFF)

S7 Fig. Pearson clusters of small genes lose separability with respect to their fPC scores. The figure shows the fPC scores ζ of small genes (<1000 bp) of all conditions coloured with respect to the *all-gene* Pearson clustering. Blue and orange indicate each one group, the dashed line symbolises the best linear separation using a SVM. We removed the linear boundary in plots where it went through the periphery instead of dividing the data points. The x-axis represents the score of the first fPC ζ^1 , the y-axis gives the score for the second fPC ζ^2 . All axes are scaled to the same size; shapes are therefore comparable. (TIFF)

S8 Fig. The small-gene fPC effect in *chd1* Δ *rsc8* **strains.** The double mutant seemingly reestablishes gene boundaries, and coordinated phasing is at least weakened after the +2 nucleosome (+1 in turquoise, +4 in blue, +6 in orange). This is true despite the fact that the A and B replicate differ. Figs (A) and (B) show the clusters for replicate A and B, and Figs (C) and (D) display their fPCs. We removed the separating boundaries in (A) and (B) because they did not reasonably divide the clusters. Nevertheless, we kept the estimated linear function in the legend to allow a comparison with other boundaries. Of particular note is the bias, which differs largely from large-gene clusters. The dashed black lines, the solid purple, and the solid green lines indicate the mean, a positive contribution, and a negative contribution, respectively.



S9 Fig. Cluster significance test for large genes. We compare the inter-cluster correlation between the Pearson clusters (magenta) and random partitioning (orange). We define the inter-cluster correlation as the distribution over all absolute pairwise Pearson indices between two genes from different clusters. It is expected that the inter-cluster correlation for the grouping determined by the *k*-mean tends to be significantly lower than the random clustering. This is verified by a KS test over 500 randomly sampled inter-cluster indices for the random and *k*-mean grouping, respectively. As the sub-sampled Pearson indices for both approaches can be sometimes fairly similar, we repeated the significance test over 500 random groupings. Left: The inter-cluster correlation distribution for the determined gene groups tends to be lower than the random clustering for all strains. Right: This notion is confirmed by the cumulative distribution function (CDF), where the Pearson CDF increases more quickly than the one for random clustering. (TIFF)

S10 Fig. Pearson clusters of large genes are linearly separable with respect to their fPC scores (replicate A). The figure shows the fPC scores ζ of all conditions coloured with respect to the Pearson clustering using only large genes (≥ 1000 bp). Blue and orange indicate each one group, the dashed line symbolises the best linear separation using a SVM. The x-axis represents the score of the first fPC ζ^1 , the y-axis gives the score for the second fPC ζ^2 . All axes are scaled to the same size; shapes are therefore comparable. It should be emphasised that only the absolute slope value matters and not the sign (i.e. pointing upwards or downwards). (TIFF)

S11 Fig. Pearson clusters of large genes are linearly separable with respect to their fPC scores (replicate B). The figure shows the fPC scores ζ of all conditions coloured with

respect to the Pearson clustering using only large genes (\geq 1000 bp). Blue and orange indicate each one group, the dashed line symbolises the best linear separation using a SVM. The x-axis represents the score of the first fPC ζ^1 , the y-axis gives the score for the second fPC ζ^2 . All axes are scaled to the same size; shapes are therefore comparable. It should be emphasised that only the absolute slope value matters and not the sign (i.e. pointing upwards or downwards).



S12 Fig. Example of an arbitrary mutant that only influences the MNase-seq average but not the collective behaviour of the nucleosome array. (A) We consider an arbitrary but fictional chromatin remodeler mutant that causes the depletion of the +2 nucleosome. All other nucleosomes remain unperturbed and keep their positioning. The MNase-seq profile at the +2 position consists only of random noise for all genes. (B) This has visibly a strong impact on the average distribution, as the +2 position is depleted in the arbitrary mutant (blue) in comparison with the WT (bold red line). (C) Since the descriptive variance decreased in our arbitrary mutant (i.e. variation that is not attributed to random noise), the fPC scores are slightly affected. Overall, however, the boundary slope remains fairly similar to the real WT A from which we constructed the arbitrary mutant. In fact, difference to WT A is smaller than the difference between the two biological replicates. In order to make sure we consider only mutants that affect the entire nucleosome array, we selected only strains for which the boundary slope notably changed with respect to the WT. (TIFF)

S13 Fig. Interdependence of Pearson clusters of MNase-seq profiles and other nuclear factors. The orange bar shows the ratio of cases where the nuclear factor could predict clustering, blue gives the wrongly classified ratio. Random guessing would be correct in 50% of the cases, which is given by the dashed black line. Consequently, the orange bar must exceed the dashed line to suggest interdependence. (TIFF)

S14 Fig. Clustering using WARD and an Euclidean distance metric indicates also a separation along the first two fPCs. We repeated gene grouping (WT A, large genes) with WARD as a fundamentally different clustering method on the nucleosome profiles using an Euclidean distance metric. However, contrary to the Pearson correlation, the Euclidean distance does not have an upper bound, and clustering was more sensitive to outliers. We removed 29 genes who had an absolute fPC score larger than 20 for either of the two fPCs. (A) The silhouette criterion indicates once again that when dividing genes into clusters, it is best to separate them into two groups. (B) Despite the fact that the Euclidean distance measures a different property than the Pearson correlation and although WARD functions fundamentally differently, the gene groups tend to separate into two groups along the first two fPCs. The boundary is admittedly not as neat as for the k-mean clusters in WT A, but they are comparable to other mutants in the study. It is expected that the boundary itself changes, as the Euclidean metric captures different properties. We can conclude that the separation of nucleosome profiles into different gene groups along the major two fPCs is not merely an artifact of our methodology. (TIFF)

S15 Fig. Clustering distribution using PCA and their principal components. Indeed, conventional PCA can separate the clusters for all genes in WT conditions ((A) and (B) for replicate *A* and *B*) similarly to fPCA. The two clusters are given in blue and orange. However, the two determined PCs ((C) and (D) for replicate *A* and *B*) differ slightly with respect to the fPCA

due to the independence assumption. Here, light blue and light orange indicate PC1 and PC2, respectively.

(TIFF)

S16 Fig. AT-ratio distribution with respect to the fPC scores. Whilst there is seemingly a slight correlation between Pearson coefficient clusters and AT-ratio in the *A* replicate, this is trend vanishes for the *B* replicate. In fact, both replicates might rather distribute AT-rich and AT-poor genes orthogonal to the dividing boundary. We plotted the original SVM boundary from the Pearson clusters with a dashed grey line to indicate that it was not determined using the AT content.

(TIFF)

Author Contributions

Conceptualization: Leo Zeitler, Cyril Denby Wilkes, Julie Soutourina, Arach Goldar.

Data curation: Leo Zeitler, Kévin André, Cyril Denby Wilkes, Arach Goldar.

Formal analysis: Leo Zeitler, Arach Goldar.

Funding acquisition: Julie Soutourina.

Investigation: Leo Zeitler, Kévin André, Cyril Denby Wilkes, Julie Soutourina, Arach Goldar.

Methodology: Leo Zeitler, Cyril Denby Wilkes, Arach Goldar.

Project administration: Julie Soutourina.

Supervision: Cyril Denby Wilkes, Julie Soutourina, Arach Goldar.

Validation: Leo Zeitler, Adriana Alberti, Julie Soutourina, Arach Goldar.

Visualization: Leo Zeitler, Arach Goldar.

Writing – original draft: Leo Zeitler, Adriana Alberti, Cyril Denby Wilkes, Julie Soutourina, Arach Goldar.

Writing – review & editing: Leo Zeitler, Adriana Alberti, Cyril Denby Wilkes, Julie Soutourina, Arach Goldar.

References

- Kornberg R. The location of nucleosomes in chromatin: specific or statistical? Nature. 1981; 292:579– 580. https://doi.org/10.1038/292579a0 PMID: 7254354
- Mavrich TN, Ioshikhes IP, Venters BJ, Jiang C, Tomsho LP, Qi J, et al. A barrier nucleosome model for statistical positioning of nucleosomes throughout the yeast genome. Genome research. 2008; 18 (7):1073–1083. https://doi.org/10.1101/gr.078261.108 PMID: 18550805
- Struhl K, Segal E. Determinants of nucleosome positioning. Nature structural & molecular biology. 2013; 20(3):267–273. https://doi.org/10.1038/nsmb.2506 PMID: 23463311
- Zhang Y, Moqtaderi Z, Rattner BP, Euskirchen G, Snyder M, Kadonaga JT, et al. Intrinsic histone-DNA interactions are not the major determinant of nucleosome positions in vivo. Nature structural & molecular biology. 2009; 16(8):847–852. https://doi.org/10.1038/nsmb.1636
- Vasseur P, Tonazzini S, Ziane R, Camasses A, Rando OJ, Radman-Livaja M. Dynamics of nucleosome positioning maturation following genomic replication. Cell reports. 2016; 16(10):2651–2665. <u>https://doi.org/10.1016/j.celrep.2016.07.083</u> PMID: 27568571
- Hughes AL, Jin Y, Rando OJ, Struhl K. A functional evolutionary approach to identify determinants of nucleosome positioning: a unifying model for establishing the genome-wide pattern. Molecular cell. 2012; 48(1):5–15. https://doi.org/10.1016/j.molcel.2012.07.003 PMID: 22885008

- Dekker J. GC-and AT-rich chromatin domains differ in conformation and histone modification status and are differentially modulated by Rpd3p. Genome biology. 2007; 8:1–14. https://doi.org/10.1186/gb-2007-8-6-r116 PMID: 17577398
- Marin-Gonzalez A, Vilhena J, Perez R, Moreno-Herrero F. A molecular view of DNA flexibility. Quarterly Reviews of Biophysics. 2021; 54:e8. https://doi.org/10.1017/S0033583521000068 PMID: 34225835
- Clapier CR, Iwasa J, Cairns BR, Peterson CL. Mechanisms of action and regulation of ATP-dependent chromatin-remodelling complexes. Nature reviews Molecular cell biology. 2017; 18(7):407–422. https://doi.org/10.1038/nrm.2017.26 PMID: 28512350
- Parnell TJ, Huff JT, Cairns BR. RSC regulates nucleosome positioning at Pol II genes and density at Pol III genes. The EMBO journal. 2008; 27(1):100–110. https://doi.org/10.1038/sj.emboj.7601946 PMID: 18059476
- Yague-Sanz C, Vázquez E, Sánchez M, Antequera F, Hermand D. A conserved role of the RSC chromatin remodeler in the establishment of nucleosome-depleted regions. Current Genetics. 2017; 63:187–193. https://doi.org/10.1007/s00294-016-0642-y PMID: 27558480
- Ocampo J, Chereji RV, Eriksson PR, Clark DJ. Contrasting roles of the RSC and ISW1/CHD1 chromatin remodelers in RNA polymerase II elongation and termination. Genome research. 2019; 29(3):407– 417. https://doi.org/10.1101/gr.242032.118 PMID: 30683752
- Biernat E, Kinney J, Dunlap K, Rizza C, Govind CK. The RSC complex remodels nucleosomes in transcribed coding sequences and promotes transcription in Saccharomyces cerevisiae. Genetics. 2021; 217(4):iyab021. https://doi.org/10.1093/genetics/iyab021 PMID: 33857307
- Ganguli D, Chereji RV, Iben JR, Cole HA, Clark DJ. RSC-dependent constructive and destructive interference between opposing arrays of phased nucleosomes in yeast. Genome research. 2014; 24 (10):1637–1649. https://doi.org/10.1101/gr.177014.114 PMID: 25015381
- Shukla MS, Syed SH, Montel F, Faivre-Moskalenko C, Bednar J, Travers A, et al. Remosomes: RSC generated non-mobilized particles with approximately 180 bp DNA loosely associated with the histone octamer. Proceedings of the National Academy of Sciences. 2010; 107(5):1936–1941. https://doi.org/ 10.1073/pnas.0904497107 PMID: 20080697
- Alén C, Kent NA, Jones HS, O'Sullivan J, Aranda A, Proudfoot NJ. A role for chromatin remodeling in transcriptional termination by RNA polymerase II. Molecular cell. 2002; 10(6):1441–1452. <u>https://doi.org/10.1016/S1097-2765(02)00778-5</u> PMID: 12504018
- Radman-Livaja M, Quan TK, Valenzuela L, Armstrong JA, Van Welsem T, Kim T, et al. A key role for Chd1 in histone H3 dynamics at the 3' ends of long genes in yeast. PLoS genetics. 2012; 8(7): e1002811. https://doi.org/10.1371/journal.pgen.1002811 PMID: 22807688
- Ocampo J, Chereji RV, Eriksson PR, Clark DJ. The ISW1 and CHD1 ATP-dependent chromatin remodelers compete to set nucleosome spacing in vivo. Nucleic acids research. 2016; 44(10):4625–4635. https://doi.org/10.1093/nar/gkw068 PMID: 26861626
- Gkikopoulos T, Schofield P, Singh V, Pinskaya M, Mellor J, Smolle M, et al. A role for Snf2-related nucleosome-spacing enzymes in genome-wide nucleosome organization. Science. 2011; 333 (6050):1758–1760. https://doi.org/10.1126/science.1206097 PMID: 21940898
- Tsukiyama T, Palmer J, Landel CC, Shiloach J, Wu C. Characterization of the imitation switch subfamily of ATP-dependent chromatin-remodeling factors in Saccharomyces cerevisiae. Genes & development. 1999; 13(6):686–697. https://doi.org/10.1101/gad.13.6.686 PMID: 10090725
- Whitehouse I, Rando OJ, Delrow J, Tsukiyama T. Chromatin remodelling at promoters suppresses antisense transcription. Nature. 2007; 450(7172):1031–1035. https://doi.org/10.1038/nature06391 PMID: 18075583
- 22. Yang JG, Madrid TS, Sevastopoulos E, Narlikar GJ. The chromatin-remodeling enzyme ACF is an ATP-dependent DNA length sensor that regulates nucleosome spacing. Nature structural & molecular biology. 2006; 13(12):1078–1083. https://doi.org/10.1038/nsmb1170 PMID: 17099699
- Yamada K, Frouws TD, Angst B, Fitzgerald DJ, DeLuca C, Schimmele K, et al. Structure and mechanism of the chromatin remodelling factor ISW1a. Nature. 2011; 472(7344):448–453. <u>https://doi.org/10.1038/nature09947 PMID: 21525927</u>
- Lieleg C, Ketterer P, Nuebler J, Ludwigsen J, Gerland U, Dietz H, et al. Nucleosome spacing generated by ISWI and CHD1 remodelers is constant regardless of nucleosome density. Molecular and cellular biology. 2015; 35(9):1588–1605. https://doi.org/10.1128/MCB.01070-14 PMID: 25733687
- 25. Oberbeckmann E, Niebauer V, Watanabe S, Farnung L, Moldt M, Schmid A, et al. Ruler elements in chromatin remodelers set nucleosome array spacing and phasing. Nature Communications. 2021; 12 (1):3232. https://doi.org/10.1038/s41467-021-23015-0 PMID: 34050140
- 26. Jiang C, Pugh BF. Nucleosome positioning and gene regulation: advances through genomics. Nature Reviews Genetics. 2009; 10(3):161–172. https://doi.org/10.1038/nrg2522 PMID: 19204718

- Weiner A, Hughes A, Yassour M, Rando OJ, Friedman N. High-resolution nucleosome mapping reveals transcription-dependent promoter packaging. Genome research. 2010; 20(1):90–100. https://doi.org/ 10.1101/gr.098509.109 PMID: 19846608
- Baldi S, Krebs S, Blum H, Becker PB. Genome-wide measurement of local nucleosome array regularity and spacing by nanopore sequencing. Nature structural & molecular biology. 2018; 25(9):894–901. https://doi.org/10.1038/s41594-018-0110-0 PMID: 30127356
- Koerber RT, Rhee HS, Jiang C, Pugh BF. Interaction of transcriptional regulators with specific nucleosomes across the Saccharomyces genome. Molecular cell. 2009; 35(6):889–902. https://doi.org/10. 1016/j.molcel.2009.09.011 PMID: 19782036
- Cole HA, Howard BH, Clark DJ. Activation-induced disruption of nucleosome position clusters on the coding regions of Gcn4-dependent genes extends into neighbouring genes. Nucleic acids research. 2011; 39(22):9521–9535. https://doi.org/10.1093/nar/gkr643 PMID: 21880600
- Singh AK, Schauer T, Pfaller L, Straub T, Mueller-Planitz F. The biogenesis and function of nucleosome arrays. Nature communications. 2021; 12(1):7011. https://doi.org/10.1038/s41467-021-27285-6 PMID: 34853297
- Lai B, Gao W, Cui K, Xie W, Tang Q, Jin W, et al. Principles of nucleosome organization revealed by single-cell micrococcal nuclease sequencing. Nature. 2018; 562(7726):281–285. https://doi.org/10.1038/ s41586-018-0567-3 PMID: 30258225
- Wan J, Lin J, Zack DJ, Qian J. Relating periodicity of nucleosome organization and gene regulation. Bioinformatics. 2009; 25(14):1782–1788. <u>https://doi.org/10.1093/bioinformatics/btp323</u> PMID: 19447785
- Deniz Ö, Flores O, Aldea M, Soler-López M, Orozco M. Nucleosome architecture throughout the cell cycle. Scientific reports. 2016; 6(1):1–11. https://doi.org/10.1038/srep19729 PMID: 26818620
- Vasseur P, Tonazzini S, Ziane R, Camasses A, Rando OJ, Radman-Livaja M. Dynamics of nucleosome positioning maturation following genomic replication. Cell reports. 2016; 16(10):2651–2665. <u>https://doi.org/10.1016/j.celrep.2016.07.083</u> PMID: 27568571
- Liu Y, Li Y, Carroll RJ, Wang N. Predictive functional linear models with diverging number of semiparametric single-index interactions. Journal of Econometrics. 2022; 230(2):221–239. <u>https://doi.org/10.1016/j.jeconom.2021.03.010</u> PMID: 36017081
- Karuppusami R, Antonisamy B, Premkumar PS. Functional principal component analysis for identifying the child growth pattern using longitudinal birth cohort data. BMC Medical Research Methodology. 2022; 22(1):76. https://doi.org/10.1186/s12874-022-01566-0 PMID: 35313828
- Luo L, Zhu Y, Xiong M. Smoothed functional principal component analysis for testing association of the entire allelic spectrum of genetic variation. European Journal of Human Genetics. 2013; 21(2):217–224. https://doi.org/10.1038/ejhg.2012.141 PMID: 22781089
- Park D, Morris AR, Battenhouse A, Iyer VR. Simultaneous mapping of transcript ends at single-nucleotide resolution and identification of widespread promoter-associated non-coding RNA governed by TATA elements. Nucleic acids research. 2014; 42(6):3736–3749. <u>https://doi.org/10.1093/nar/gkt1366</u> PMID: 24413663
- Klein-Brill A, Joseph-Strauss D, Appleboim A, Friedman N. Dynamics of chromatin and transcription during transient depletion of the RSC chromatin remodeling complex. Cell reports. 2019; 26(1):279– 292. https://doi.org/10.1016/j.celrep.2018.12.020 PMID: 30605682
- 41. Hebb DO. Organization of behavior. New York: Wiley and Sons J Clin Psychology. 1949; 6(3):335–307.
- Vainshtein Y, Rippe K, Teif VB. NucTools: analysis of chromatin feature occupancy profiles from highthroughput sequencing data. BMC genomics. 2017; 18:1–13. <u>https://doi.org/10.1186/s12864-017-3580-2 PMID: 28196481</u>
- Baek S, Lee I. Single-cell ATAC sequencing analysis: From data preprocessing to hypothesis generation. Computational and structural biotechnology journal. 2020; 18:1429–1439. <u>https://doi.org/10.1016/j.csbj.2020.06.012</u> PMID: 32637041
- 44. McInnes L, Healy J, Melville J. Umap: Uniform manifold approximation and projection for dimension reduction. arXiv preprint arXiv:180203426. 2018;.
- **45.** Van der Maaten L, Hinton G. Visualizing data using t-SNE. Journal of machine learning research. 2008; 9(11).
- Kubik S, Bruzzone MJ, Jacquet P, Falcone JL, Rougemont J, Shore D. Nucleosome stability distinguishes two different promoter types at all protein-coding genes in yeast. Molecular cell. 2015; 60 (3):422–434. https://doi.org/10.1016/j.molcel.2015.10.002 PMID: 26545077
- Cairns BR, Lorch Y, Li Y, Zhang M, Lacomis L, Erdjument-Bromage H, et al. RSC, an essential, abundant chromatin-remodeling complex. Cell. 1996; 87(7):1249–1260. <u>https://doi.org/10.1016/S0092-8674</u> (00)81820-6 PMID: 8980231

- Krietenstein N, Wal M, Watanabe S, Park B, Peterson CL, Pugh BF, et al. Genomic nucleosome organization reconstituted with pure proteins. Cell. 2016; 167(3):709–721. <u>https://doi.org/10.1016/j.cell.2016</u>. 09.045 PMID: 27768892
- Damelin M, Simon I, Moy TI, Wilson B, Komili S, Tempst P, et al. The genome-wide localization of Rsc9, a component of the RSC chromatin-remodeling complex, changes in response to stress. Molecular cell. 2002; 9(3):563–573. https://doi.org/10.1016/S1097-2765(02)00475-6 PMID: 11931764
- Ng HH, Robert F, Young RA, Struhl K. Genome-wide location and regulated recruitment of the RSC nucleosome-remodeling complex. Genes & development. 2002; 16(7):806–819. <u>https://doi.org/10.1101/gad.978902</u> PMID: 11937489
- Quan TK, Hartzog GA. Histone H3K4 and K36 methylation, Chd1 and Rpd3S oppose the functions of Saccharomyces cerevisiae Spt4–Spt5 in transcription. Genetics. 2010; 184(2):321–334. https://doi.org/ 10.1534/genetics.109.111526 PMID: 19948887
- Park D, Shivram H, Iyer VR. Chd1 co-localizes with early transcription elongation factors independently of H3K36 methylation and releases stalled RNA polymerase II at introns. Epigenetics & chromatin. 2014; 7(1):1–11. https://doi.org/10.1186/1756-8935-7-32 PMID: 25395991
- Simic R, Lindstrom DL, Tran HG, Roinick KL, Costa PJ, Johnson AD, et al. Chromatin remodeling protein Chd1 interacts with transcription elongation factors and localizes to transcribed genes. The EMBO journal. 2003; 22(8):1846–1856. https://doi.org/10.1093/emboj/cdg179 PMID: 12682017
- 54. Marfella CG, Imbalzano AN. The Chd family of chromatin remodelers. Mutation Research/Fundamental and Molecular Mechanisms of Mutagenesis. 2007; 618(1-2):30–40. https://doi.org/10.1016/j.mrfmmm. 2006.07.012 PMID: 17350655
- Yen K, Vinayachandran V, Batta K, Koerber RT, Pugh BF. Genome-wide nucleosome specificity and directionality of chromatin remodelers. Cell. 2012; 149(7):1461–1473. <u>https://doi.org/10.1016/j.cell.</u> 2012.04.036 PMID: 22726434
- André KM, Aiach NG, Martinez-Fernandez V, Zeitler L, Alberti A, Goldar A, et al. Functional interplay between Mediator and RSC chromatin remodeling complex controls nucleosome-depleted region maintenance at promoters. Cell Reports. 2023; 42(5). PMID: 37133993
- 57. Krueger F. Trim Galore; 2019. https://github.com/FelixKrueger/TrimGalore/releases/tag/0.6.5.
- Martin M. Cutadapt removes adapter sequences from high-throughput sequencing reads. EMBnet journal. 2011; 17(1):10–12. https://doi.org/10.14806/ej.17.1.200
- Langmead B, Salzberg SL. Fast gapped-read alignment with Bowtie 2. Nature methods. 2012; 9 (4):357–359. https://doi.org/10.1038/nmeth.1923 PMID: 22388286
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The sequence alignment/map format and SAMtools. bioinformatics. 2009; 25(16):2078–2079. https://doi.org/10.1093/bioinformatics/btp352 PMID: 19505943
- Ramírez F, Ryan DP, Grüning B, Bhardwaj V, Kilpert F, Richter AS, et al. deepTools2: a next generation web server for deep-sequencing data analysis. Nucleic acids research. 2016; 44(W1):W160–W165. https://doi.org/10.1093/nar/gkw257 PMID: 27079975
- Georges A, Gopaul D, Denby Wilkes C, Giordanengo Aiach N, Novikova E, Barrault MB, et al. Functional interplay between Mediator and RNA polymerase II in Rad2/XPG loading to the chromatin. Nucleic acids research. 2019; 47(17):8988–9004. https://doi.org/10.1093/nar/gkz598 PMID: 31299084
- 63. Kaufman L, Rousseeuw PJ. Finding groups in data: an introduction to cluster analysis. John Wiley & Sons; 2009.
- Rousseeuw PJ. Silhouettes: a graphical aid to the interpretation and validation of cluster analysis. Journal of computational and applied mathematics. 1987; 20:53–65. <u>https://doi.org/10.1016/0377-0427(87)</u> 90125-7
- Ramos-Carreño C, Torrecilla JL, Carbajo-Berrocal M, Marcos P, Suárez A. scikit-fda: a Python package for functional data analysis. arXiv preprint arXiv:221102566. 2022;.